

Analysing the implementation of a material requirements planning (MRP) system into an engineer-to-order (ETO) company: the case of National Oilwell Varco Norway (NOVN)

Elisabeth Fidjeland & Tone Skovly Gabrielsen

**Supervisors** Tina Comes and Naima Saeed

This master's thesis is carried out as a part of the education at the University of Agder and is therefore approved as a part of this education. However, this does not imply that the University answers for the methods that are used or the conclusions that are drawn.

University of Agder, 2014

School of Business and Law

# PREFACE

This master thesis is written by Elisabeth Fidjeland and Tone Gabrielsen in 2014 and represents the concluding work of the finalisation of our Master's degree in Business Administration with a major in Performance Analysis and Project Management at the School of Business and Law, University of Agder.

The purpose of the thesis is to investigate if implementation of a material requirements planning (MRP) system supports the operations strategy of an engineer-to-order (ETO) company. The research was conducted by performing a case study of National Oilwell Varco Norway (NOVN).

The scope and the content of the thesis have been developed by the authors in cooperation with Lars Lohne at NOVN. The authors' supervisors, Tina Comes and Naima Saeed, at the University of Agder has also contributed with their advice.

The authors would like to thank NOVN for the opportunity to collaborate with them, as well as express their sincere gratitude to Lars Lohne, their contact person at NOVN, for his support and help in identifying the topic of the master thesis, as well as his advices, comments, information providing, and insights throughout the research.

The authors would also like to express their deepest appreciation to their supervisors, Tina Comes and Naima Saeed, for their good and valuable comments and guidance throughout the work on the thesis and before the thesis was handed in, respectively.

The authors would also like to thank several key informants at NOVN for participating in the interviews and for making the case study possible. And last, but not least, the authors would like to thank those NOVN employees that have helped both with inquiries and with gathering and structuring of data, as well as those who have proofread the master thesis.

Kristiansand, 29.10.2014

Elisabeth Fidjeland

Tone Skovly Gabrielsen

## ABSTRACT

A material requirements planning (MRP) system is a computer-based planning and control system whose main objectives are to provide the right part at the right time, and to meet the schedules for completed products. The development of these systems revolutionised the manufacturing industry, and lead to it being adopted by many companies. The expectations of the systems were high, both from academia and industry in the subject area of production planning and control. However, the widespread use of the system has uncovered several failures, mainly because the systems are implemented under the assumption that "one-size-fits-all", and thus do not differentiate between various operations strategies. Prior research has already identified MRP systems as successful production planning and control systems in several operations strategies. Despite its importance, the previous research on MRP systems has not thoroughly addressed the systems strategic fit with an engineer-to-order (ETO) operations strategy. This thesis therefore focuses on the use of an MRP system in an ETO environment, and the overall objective is to investigate if implementation of an MRP system supports the operations strategy of an ETO company.

To help investigate the overall objective, a literature review and a case study has been conducted. The literature review was carried out to provide a theoretical base for the research and a foundation for the future work of the research. A case study was conducted to help get a better understanding of an MRP system's strategic fit in an ETO company to draw parallels between theory and practice. Numerical data has been collected to conduct statistical analysis. The case study company is a large ETO company that is about to implement an MRP system and that previously have used a similar system in some of its departments. Qualitative data from the case study have mainly been conducted through interviews and informal conversations with key informants employed in the case study company.

The result of this research shows that there is a clear misalignment between the decision support provided by an MRP system and the decision support required by an ETO company. The product-, market- and process characteristics of an ETO company are too much of a constraining factor for the MRP system, which may lead to reduced competitiveness. Furthermore, the research suggests that organisational factors, such as education level of employees, company size and culture have significant impacts on implementation of an MRP system.

The results gathered from the research have a foundation from relevant theory, which strengthens the quality of the thesis. The thesis has therefore contributed with increased knowledge and provides a better understanding of the use of an MRP system in an ETO company. In particular the definitions in the thesis, the identified variables, and the frameworks should be of interest for researchers, management, and consultant in the area of production planning and control (PPC). The research also has important implications for top management and policy makers in implementing an MRP system, as these stakeholders need to communicate effectively with their organisation about their MRP adoption intentions.

Case study findings suggest that MRP systems are not suitable for ETO products, and that MRP implementation is influenced by, but not necessarily bound by, existing national and organisational factors. The findings of this study aid the management of organisations that are implementing MRP systems to gain a better understanding of the likely challenges they may face and enables them to put in place appropriate measures to mitigate the risk of implementation failures.

# **ABBREVIATIONS**

ANOVA	Analysis of Variance			
АТО	Assemble-to-Order			
BOM	Bill of Material			
вто	Buy-to-Order			
CODP	Customer Order Decoupling Point			
EOQ	Economic Order Quantity			
ER	Engineer Responsible			
ERP	Enterprise Resource Planning			
ЕТО	Engineer-to-Order			
JIT	Just-in-Time			
LFL	Lot-for-lot			
MPS	Master Production Schedule			
MRP	Material Requirements Planning			
MRPII	Manufacturing Resource Planning			
МТО	Make-to-Order			
MTS	Make-to-Stock			
NOV	National Oilwell Varco			
NOVN	National Oilwell Varco Norway			
OM	Operations Management			
PLC	Product life cycle			
PPB	Part Period Balancing			
PPC	Production Planning and Control			
SME	Small and Medium sized Enterprises			
SPSS	Statistical Package for the Social Sciences			
STS	Ship-to-Stock			

# **LIST OF CONTENT**

<u>PREI</u>	FACE	2
ABS]	<u> </u>	3
		_
ABR	REVIATIONS	<u>5</u>
LIST	OF CONTENT	6
<u>LIST</u>	OF FIGURES	8
<u>LIST</u>	OF TABLES	8
1 I	NTRODUCTION	9
<u> </u>		4.0
<u>2</u> 1	THEORETICAL BACKGROUND	<u>12</u>
2.1	<b>O</b> PERATIONS MANAGEMENT	12
2.1.1	PLANNING AND CONTROL	13
2.1.2	OPERATIONS STRATEGY	15
2.1.3	DESIGN	17
2.2	STRATEGIC FIT	24
2.3	IMPACT OF PPC SYSTEMS	25
2.4	MATERIAL REQUIREMENTS PLANNING	29
2.4.1	MATERIAL REQUIREMENTS PLANNING	29
2.4.2	EVOLUTION OF MRP	29
2.4.3	BASIC CONCEPTS OF MRP	29
2.4.4	PREREQUISITES AND ASSUMPTIONS OF MRP	
2.4.5	MRP ADVANTAGES	
2.4.6	MRP DRAWBACKS	
2.4.7	FACTORS AFFECTING THE SUCCESSFUL IMPLEMENTATION OF MRP SYSTEMS	
2.4.8	LITERATURE REVIEW ON THE USE OF AN MIRP SYSTEM IN AN ETO ENVIRONMENT	
<u>3</u> F	RESEARCH OBJECTIVE AND RESEARCH QUESTIONS	40
3.1	RESEARCH OBJECTIVE	40
3.2	Research questions	40
4 F	RESEARCH METHODOLOGY	45
4.1	Research design	45
4.2	Research methods	46
4.2.1	LITERATURE REVIEW	47
4.2.2	CASE STUDY	47
4.3	SAMPLING	49
4.4	DATA COLLECTION AND ANALYSIS	49
4.4.1	DATA COLLECTION	49
4.4.2	DATA ANALYSIS	50
4.5	QUALITY OF THE RESEARCH DESIGN	51
4.5.1	EXTERNAL VALIDITY	51
4.5.2	INTERNAL VALIDITY	52
4.5.3	CONSTRUCT VALIDITY	
4.5.4	KELIABILITY	53
4.6	SUMMARY	53

5.1 NATIONAL OILWELL VARCO INC. 54   5.1.1 NATIONAL OILWELL VARCO NORWAY 54   5.2 ROUGHNECK UNIT. 57   5.2.1 ASSEMBLE-TO-ORDER OPERATIONS STRATEGY 58   5.2.2 THE ATO BASED PPC SYSTEM 56   6 RQ1: EVALUATION OF THE ATO BASED PPC SYSTEM 67   6.1 QUANTITATIVE AND QUALITATIVE EVALUATION 67   6.2 QUANTITATIVE MEASURES 67   6.2.1 QUALITY 67   6.2.2 SPEED 64   6.2.3 DEPENDABILITY 67
5.1.1 NATIONAL OILWELL VARCO NORWAY 54   5.2 ROUGHNECK UNIT 57   5.2.1 ASSEMBLE-TO-ORDER OPERATIONS STRATEGY 53   5.2.2 THE ATO BASED PPC SYSTEM 53   6 RQ1: EVALUATION OF THE ATO BASED PPC SYSTEM 67   6.1 QUANTITATIVE AND QUALITATIVE EVALUATION 67   6.2 QUANTITATIVE MEASURES 67   6.2.1 QUALITY 67   6.2.2 SPEED 64   6.2.3 DEPENDABILITY 67
5.2 ROUGHNECK UNIT
5.2.1 ASSEMBLE-TO-ORDER OPERATIONS STRATEGY 53   5.2.2 THE ATO BASED PPC SYSTEM 56   6 RQ1: EVALUATION OF THE ATO BASED PPC SYSTEM 67   6.1 QUANTITATIVE AND QUALITATIVE EVALUATION 67   6.2 QUANTITATIVE MEASURES 67   6.2.1 QUALITY 67   6.2.2 SPEED 64   6.2.3 DEPENDABILITY 67
5.2.2 THE ATO BASED PPC SYSTEM 59   6 RQ1: EVALUATION OF THE ATO BASED PPC SYSTEM 67   6.1 QUANTITATIVE AND QUALITATIVE EVALUATION 67   6.2 QUANTITATIVE MEASURES 67   6.2.1 QUALITY 67   6.2.2 SPEED 64   6.2.3 DEPENDABILITY 67
6 RQ1: EVALUATION OF THE ATO BASED PPC SYSTEM 67   6.1 QUANTITATIVE AND QUALITATIVE EVALUATION 67   6.2 QUANTITATIVE MEASURES 67   6.2.1 QUALITY 67   6.2.2 SPEED 64   6.2.3 DEPENDABILITY 67
6.1 QUANTITATIVE AND QUALITATIVE EVALUATION 67   6.2 QUANTITATIVE MEASURES 67   6.2.1 QUALITY 67   6.2.2 SPEED 64   6.2.3 DEPENDABILITY 67
6.2 QUANTITATIVE MEASURES
6.2.1 QUALITY 63   6.2.2 SPEED 64   6.2.3 DEPENDABILITY 67
6.2.2 Speed
6.2.3 DEPENDABILITY
6.2.4 FLEXIBILITY
6.2.5 COST
6.2.6 VALUE ADDED
6.3 QUALITATIVE MEASURES
6.3.1 EDUCATION OF PERSONNEL
6.3.2 GOALS
6.3.3 SUPPORT
6.3.4 GAINING ACCEPTANCE
7 RQ2: MRP IN AN ETO ENVIRONMENT
8 ADDITIONAL ELEMENTS TO CONSIDER WHEN IMPLEMENTING AN MRP SYSTEM
8.1 CULTURES INFLUENCE ON MRP IMPLEMENTATION
8.2 THE INFLUENCE OF CULTURE ON THE IMPLEMENTATION OF AN MRP SYSTEM IN NOVN
8.3 OTHER ASPECTS THAT NOVN MUST TAKE INTO CONSIDERATION WHEN IMPLEMENTING THE MRP
SYSTEM
<u>9</u> <u>CONCLUSION</u>
9.1 Answers to the research ouestions
9.1.1 RO1: WHAT IS THE IMPACT OF AN ATO BASED PPC SYSTEM ON AN ETO COMPANY?
9.1.2 RO2: How can an MRP system be used to support the PPC needs of an ETO company?90
9.1.3 SUMMARY OF RESEARCH FINDINGS IN RELATION TO THE OVERALL RESEARCH OBJECTIVE OF THE THESI
97
9.2 GENERALIZABILITY
9.3 SCOPE AND LIMITATIONS
9.4 FURTHER RESEARCH
<u>REFERENCES</u>
APPENDIX

# **LIST OF FIGURES**

Figure 1.1: Thesis structure	11
Figure 2.1: A general model of operations management and operations strategy	13
Figure 2.2: The product life cycle	17
Figure 2.3: Different operations strategies related to different CODP	19
Figure 2.4: Finding the zone of strategic fit	25
Figure 2.5: The basic MRP system	30
Figure 3.1: Research model	44
Figure 6.1: Research model	62
Figure 6.2: Research model with significant impacts	77

# LIST OF TABLES

Table 2.1: The tangible and subjective performance measures	28
Table 2.2: A summary of key MRP studies of relevance to an ETO context	36
Table 4.1: An overview of the research methodology used in the thesis	53

## **1** INTRODUCTION

The globalisation of the economy, increased availability of information and advancement in information technology has led to increasing competiveness in today's marketplace. In the past decades, the main focus of competition has shifted from price towards faster and more reliable delivery as well as the need for manufacturing firms to become more responsive to the increasing changes in customer- and market requirements (Hicks et al., 2001). One of the key factors of manufacturing responsiveness and success is a company's planning and control of material flow (Little et al., 2000).

The earliest contributions to the development and advancement of material planning and control systems are dated back to the early 20th century. However, the most significant contributions have been made the last five decades (Mabert, 2007). The development of the computer-based manufacturing management system MRP during the 1960s revolutionised the manufacturing industry (Islam et al., 2013). Academia and the manufacturing industry have since then put a great deal of effort into further developing the system. Today, MRP is the most popular manufacturing management system in the world (Sum & Ng, 2014).

Like other supply chain management systems, MRP systems have been rapidly implemented under a "one-size-fits-all" mind-set (Gosling et al., 2013). However, the widespread use of the system has unveiled that behind the large success of the MRP system, there are also many failures (J. C. Anderson et al., 1984). For instance, even though MRP systems perform well with many different operations strategies, they do not, at the present date, perform satisfactorily in companies with an ETO operations strategy (Bertrand & Muntslag, 1993; Little et al., 2000).

An ETO operations strategy is based on demand driven practices to fulfil specific customer orders (Jin & Thomson, 2003). The main characteristics of environments with this type of operations strategy are high levels of customisation for each product and that the products are typically managed in a project environment. Another important characteristic is that production flow is driven by actual customer orders (Gosling et al., 2013). In an ETO operations strategy the decoupling point, which is the point in the value chain where the product is linked to a specific customer order, is located at the design stage (J. Olhager, 2012).

However, as there is a lack of order specifications at the time of an order placement and continuous changes must be accommodated as a product gets finalised. The location of the decoupling point in an ETO environment makes manufacturing planning therefore complex (Islam et al., 2013).

For an operations strategy to work optimally in various business environments, problems need to be structurally the same. However, as the various operations strategies have distinct characteristics leading to structurally different problems, the "one-size-fits-all" mind-set regarding MRP is not correct. The fact that companies in various business environments apply MRP systems indicates that there is a gap between theory and practice (Little et al., 2000). And even though there exist extensive literature that has reviewed MRP systems in various operations strategies, the authors of this master thesis has had difficulties finding literature that touches upon the fit between an MRP system and an ETO operations strategy. It is therefore obvious that for a company to meet today's market requirements and stay competitive; there is a great need to understand the fit between the ETO operations strategy and the capabilities of an MRP system.

Based on this background, this master thesis reports on a qualitative and quantitative research to make a theoretical contribution to whether implementation of an MRP system supports the operations strategy of an ETO company. This has been done by working with a specific case, National Oilwell Varco Norway (NOVN), which is going to implement an MRP system in the near future. To narrow the gap between theory and practice, practical relevance and realism has guided the objectives, scope and limitations for this research, as well as the applied research methodology, which will be thoroughly elaborated in Chapter 4. Both ETO and MRP will be further elaborated in Chapter 3.

The structure of the thesis can be visually followed using the schematic in Figure 1.1. The figure shows that Chapters 2, 3, and 4 are theoretical chapters covering the main theory relevant for conducting the research. Chapters 5, 6, 7 and 8 present the quantitative and qualitative findings from the case study phase of the research using data collected from NOVN and data gathered from interviews. Chapter 8 also includes a section where the findings of the different research phases are integrated. Chapter 9 draws conclusions from all phases of the research.



Figure 1.1: Thesis structure

### **2 THEORETICAL BACKGROUND**

The aim of this chapter is to establish the theoretical foundation of the thesis to be able to investigate what the impact will be on the performance of an ETO company when implementing an MRP system. This will be done by establishing the role of the production planning and control (PPC) system within operations management, the PPC systems relation to operations strategy, what factors should guide the choice of PPC approaches and how PPC performance can be evaluated. This chapter therefore provides an overview of the literature related to the main research areas within operations management (OM) that are relevant for this research.

Section 2.1 gives a thorough review about OM and its key responsibilities and relations, which are the basis for further structuring the theory related to PPC. Sub-section 2.1.1 introduces planning and control and outlines PPC's importance for a company's performance. Sub-section 2.1.2 elaborates on operations strategy and its perspectives, which have an impact on PPC. Sub-section 2.1.3 introduces approaches for designing PPC and focuses on the ETO operations strategy and its determinants, as it is applicable in this research. Section 2.2 outlines the importance of a PPC system and a company's long-term success of linking corporate objectives, operations and operations strategy together. Section 2.3 identifies measures for evaluating PPC performance, which is tested empirically in Chapter 6. Section 2.4 concludes the chapter with a review of MRP, one of the major topics of this research.

#### 2.1 Operations management

The topic of this thesis focus on MRP, ETO and PPC, and both Jonsson (2008, p. 46) and Slack et al. (2004, p. 6) view these areas as being a part of operations management (OM). There are several similar definitions of OM. Heizer and Render defines OM as "activities that relate to the creation of goods and services through the transformation of inputs into outputs" (Heizer & Render, 1988, p. 4). Whilst Slack et al. (2004, p. 6) defines OM as the "the term used for the activities, decisions and responsibilities of operations managers" (Slack et al., 2004, p. 6). However, due to this research's objectives and aims, and because this research uses a large company as its case study, Barnes (2008, p. 3) definition of OM has been found to be applicable. Barnes (2008) defines OM as "the management of the resources and processes required by an organisation to produce goods or services for customers" (Barnes, 2008, p. 3). Figure 2.1 displays OM's key responsibilities and relations, which will be explained in the following.



Figure 2.1: A general model of operations management and operations strategy (Slack et al., 2004, p. 31).

#### 2.1.1 Planning and control

Planning and control is important for all operations processes, both within and across companies, and is concerned with managing the operations processes and ensuring that they run effectively and efficiently (Klassen & Menor, 2006, p. 135; Slack et al., 2004, p. 322). While planning is a formalisation of what is intended to happen in the future, control is the process of monitoring and revising plans if changes occur to achieve the objectives of the original plan (Slack et al., 2004, pp. 324-325).

Planning and control involves multiple activities. However, four activities are considered to be most common (Slack et al., 2004, pp. 333-346):

- Loading determining the amount of work allocated to a work centre.
- Sequencing determining the order in which the work will be tackled.
- Scheduling determining at what time various work should start and end.
- Monitoring and control ensuring that planned activities are happening and rectify deviations through intervention.

These activities are required to reconcile supply and demand in adherence to customer requirements (Slack et al., 2004, p. 333).

The balance between planning and control changes during the product life cycle. In the longterm (month or years), the emphasis is on planning and the main focus is on the financial objectives. The specific customer needs are still unknown, thus a company can only plan on an aggregated level. In the medium-term (week or months), plans get more detailed, and operational objectives complement the financial ones. On the short-term (hours and days), control is of paramount importance to address the customers' needs as they are articulated. At this stage it is difficult to make planning changes, as resources are already set. Changes will therefore be made based on a company's prioritisation of their performance measures (Slack et al., 2004, pp. 325-326).

#### 2.1.1.1 Production planning and control systems

To execute planning and control, companies apply PPC systems. Due to globalisation and increased competition together with rising customer expectations, more volatile planning horizons, and shifts in demand, planning and control has become more complicated in the past decades (Grant, 2003). Shifts in demand require shorter planning horizons, thus creating a higher need for control. Because of this, PPC systems have become increasingly important (Stevenson et al., 2005).

A PPC system should support decision makers by providing an overview of key planning and control activities and by defining data and information that managers need (Vollmann et al., 2005, p. 4). Typical functions of a PPC system are material requirements planning, demand management, capacity planning, scheduling, and sequencing. The purpose of these activities are to reduce Work in Progress (materials and partly finished products that are at various stages of the production process) reduce lead-times, lower stock holding costs and improve responsiveness (Stevenson et al., 2005). The amount and usage of PPC systems are widespread (J. C. Anderson et al., 1984; Schroeder et al., 1981; Stevenson et al., 2005). The most commonly used PPC systems today are material requirements planning (MRP), manufacturing resource planning (MRPII), just in time (JIT), Kanban and workload control (Zijm, 2000).

The choice of PPC is closely linked to operations strategy (Hayes, 1985; J. Olhager, 2003a; Porter, 1983, 1985; Stevenson et al., 2005). However, there are rival opinions regarding whether PPC systems should be defining the operations strategy of the company, or if the operations strategy should be guiding the choice of PPC systems. Skinner (2007) Porter (1983), Stevenson et al. (2005) and Slack et al. (2004, p. 65) explain that to drive competitiveness in companies, operations strategy should drive the choice of PPC system. Hayes (1985) on the other hand argues that technology is the foundation of operations strategy and that the PPC system should therefore guide the choice of operations strategy. In line with the wording of the overall research objective of this thesis, the authors adopt Porters (1983), Skinner (2007), Stevenson et al. (2005) and Slack et al. (2004, p. 65) view as the rival opinion above suggests that an operations strategy and a PPC systems will not achieve a company's full potential unless there exist a link between them. This further outlines that the choice of a PPC system is a strategic decision and critical to the success of any company (J. Olhager, 2003a; Stevenson et al., 2005).

There exist a wide range of PPC systems, and they all claim to be universally suitable, making it difficult to choose the most appropriate one (Stevenson et al., 2005). Stevenson et al. (2005) argues that the systems claim to be universally suitable because there exist no indications of what type of companies that can benefit from the systems. Thus, such systems will not be able to cater all specific PPC needs, as each production environment and company has its unique characteristics (Bertrand & Muntslag, 1993; Stevenson et al., 2005). Slack et al. (2004, p. 324) explains that a company's constraints and capabilities are defined within their PPC system, and the PPC system thus serves as the interface between a company's strategy and operations. The company's operations strategy and the design of the company's operations therefore have important implications for the appropriateness of a PPC system (cf. Figure 2.1). In the following sections the different factors that should guide the choice of PPC approaches will be introduced and explained. These factors will be used as a tool for structuring qualitative and quantitative data in Chapter 6.

#### 2.1.2 Operations strategy

There are several definitions of operations strategies. Swamidass and Newell (1987) describe operations strategy as a tool for use of manufacturing strengths as a competitive weapon to attain business and corporate objectives. Platts et al. (1998, p. 517) have defined operations strategy as "a pattern of decisions, which determine the capability of a manufacturing system

and specify how it will operate, in order to meet a set of manufacturing objectives which are consistent with the overall business objectives". Skinner (2007) says that operations strategy can be seen as a set of manufacturing policies designed to maximise performance of trade-offs among success criteria to meet the manufacturing task determined by a corporate strategy. However, this research argues that an operations strategy can be seen as a company's plan of action, which describes the company's pattern of strategic decision and actions as well as how the company will use their resources in operations and how these resources can be a source of competitive advantage (Barnes, 2008, p. 28; K. K. Boyer & McDermott, 1999; Hill & Hill, 2009, p. 31; Slack et al., 2004, p. 67). In the following sections, a number of perspectives that has continual influence on operations strategy, and hence encompasses the choice of PPC system, will be introduced and explained (Slack et al., 2004, pp. 67,70).

#### 2.1.2.1 Company objectives and performance measures

The objective of a company is concerned with what the company would like to achieve. It defines the overall strategic direction of the company as a whole as well as which market segments it intends to serve and grow in (Barnes, 2008, p. 24; Hill & Hill, 2009, p. 24; Slack et al., 2004, p. 68). The ultimate objective of implementing a PPC system is to contribute to the profits of the company. This is accomplished by keeping the customer satisfied by meeting delivery schedules. More specific, the objectives of a PPC system are to establish schedules and routines for work, which will ensure the optimum utilisation of materials, workers and machines, and to provide the means for ensuring the operation of the company in accordance with these plans (Nagare, 2007).

Performance measurements can be seen as measurement of outcomes and results, which are used to assess how well a company is achieving its desired objectives. Several authors have defined five performance measures for operations within a company. These are: quality, speed, dependability, flexibility and costs, which all have an impact on customer satisfaction (Ferdows & Meyer, 1990; Flynn & Flynn, 2004; Gunasekaran et al., 2004; Slack et al., 2004, p. 44; Vokurka et al., 2002; G. P. White, 1996). It is unlikely that a company can excel simultaneously on every aspect (Slack et al., 2004, p. 44). The importance of the measures should therefore reflect which measures are most valued by a company's customers, and the company should prioritise the measures accordingly. For example, if a particular market segments objective is to produce at low cost, a company will strive to achieve an operations strategy that facilitates producing at low cost and in high volumes, leading to speedy

deliveries and low flexibility (Barnes, 2008, p. 25).

#### 2.1.2.2 The product life cycle

Products require different operations management in each stage of the product life cycle (PLC) (Slack et al., 2004, pp. 75-76). The PLC is divided into four stages; introduction, growth, maturity, and decline (cf. Figure 2.2), and can be described as the period of time where a product is developed, brought to the market and eventually removed from the market (C. R. Anderson & Zeithaml, 1984; Jonsson, 2008, p. 159; Rink, 1979; Slack et al., 2004, p. 76). Thus the PLC represents the time when it is financially defensible to sell a product in the market (Jonsson, 2008, p. 159). For example, a product in the introduction stage is often innovative and thus few competitors are offering the same product. Slow growth and uncertainty is predominant, as the customer's needs are not yet fully understood. For the company to stay competitive, its operations must be flexible enough to accommodate changes in customer needs. In contrast, when the product reaches its maturity stage, the focus is on reducing cost and achieving dependable supplies, as fewer but larger competitors dominate the market (C. R. Anderson & Zeithaml, 1984; Slack et al., 2004, p. 76).



Figure 2.2: The product life cycle. The figure is adapted from Slack et al. (2004, p. 76).

#### 2.1.3 Design

The design of a company's product and processes, as well as the market requirements, impose restrictions on operations strategy (cf. Figure 2.1) (W. L. Berry & Terry, 1992; Stevenson et al., 2005). The following sections investigate the role of the customer order decoupling point

(CODP) in several operations strategies and processes, in addition to establishing the fundamental difference of operations relative to the CODP (Bertrand & Muntslag, 1993).

#### 2.1.3.1 Operations strategies

Various literatures propose several operations strategies that can be employed by companies. Based on Naylor et al. (1999), Yang and Burns (2003), Olhager (2003b), and Wikner and Rudberg (2005) six different operations strategies are defined: engineer-to-order (ETO), buy-to-order (BTO), make-to-order (MTO), assemble-to-order (ATO), make-to-stock (MTS), and ship-to-stock (STS). However, Olhager (2003) simplified the classification of operations strategies to include MTS, MTO, ATO, and ETO.

- *Make-to-stock* (MTS) is an operations strategy that forecasts demand to determine how much should be produced of a certain item. The strategy converts components into finished end-items, and awaits customer orders. This strategy relies heavily on accuracy of the demand forecasts.
- *Make-to-order* (MTO) is an operations strategy that only manufactures the end product after the customer places an order. The strategy allows customers to purchase products that are customised to their preferences, and the products usually consist of both standard items and custom-made items. This strategy resolves the problem of excessive inventory, which is common within the MTS strategy.
- Assemble-to-order (ATO) is an operations strategy where the main components and subassemblies of a specific product are produced to stock based on a forecast of demand. The remaining components and subassemblies are not completed until the customer orders a product and includes its specific requirements, thus the ATO strategy can be seen as a hybrid of MTS and MTO. It is therefore a strategy that efficiently delivers a high level of product variety to customers, while maintaining reasonable response time and costs.
- *Engineer-to-order* (ETO) is an operations strategy that is based on demand driven practices where components are designed, engineered, and then built to fit specific customer specifications in response to customer orders. ETO will be further elaborated in Section 2.1.3.1.3.

(Bertrand & Muntslag, 1993; Hallgren & Olhager, 2006; J. Olhager, 2003b)

#### 2.1.3.1.1 Customer order decoupling point

The location of the customer order decoupling point (CODP) differentiates the operations strategies. It therefore has a key role in the design and managing of supply chains (Jodlbauer et al., 2012). Olhager (2003) defined the CODP as "the specific point in the manufacturing value chain for a product, where the product is linked to a specific customer order" (J. Olhager, 2003b, p. 320).

The four different operations strategies are related to the ability to accommodate customisation or the size of the product range, and their CODP's therefore have different positions. This can be seen in Figure 2.3, where the dotted lines represent the production activities that are forecast driven, and the solid lines represent the customer-order-driven activities.

Product delivery strategy	Design	Fabrication & procurement	Final assembly	Shipment
Make-to-stock			••••••	CODP
Assemble-to-order			CODP	
Make-to-order		CODP ———		
Engineer-to-order	CODP ———			

Figure 2.3: Different operations strategies related to different customer order decoupling points. The figure is adapted from Olhager (2003b).

As the CODP is the transition from customer-independent production to customer-dependent production, all activities before the CODP is based on the company's market research and planning, and all activities after the CODP is based on the actual customer's order.

#### 2.1.3.1.2 Determinants of operations strategy

The four identified operations strategies have different characteristics that enable a company to meet the customer's individual requirements. The location of the CODP should therefore be strategically motivated (J. Olhager, 2003b). Various literature propose several determinants of operations strategies, but there is a consensus that the placement of the CODP is mainly affected by a products demand and characteristics, as well as its belonging

processes (Marshall, 1997; van Donk, 2003). Olhager (2003b) divides the most important characteristics into three categories related to market, product, and production. These will be further elaborated in association with ETO as an adequate understanding of the characteristics of an ETO supply chain is necessary for establishing the link between the PPC system and the PPC needs of an ETO company (W. L. Berry & Terry, 1992).

#### 2.1.3.1.3 Engineer-to-order

ETO products tend to be complex, technical and highly specialised. Because of this, ETO companies usually specialise in one type of product or related types of products, and thus have their own core technical competency. Sectors that usually fall into the ETO environment are construction, shipbuilding, complex sub-assemblies, and capital goods, which include machinery and oil exploration rig (Rahim & Baksh, 2003). Because of this, the drilling supply sector falls under the ETO operations strategy.

Authors use different terms when describing the ETO environment, which show that there is a lack of clarity of what is the appropriate terminology to use. For instance, project (Elfving et al., 2005), one-of-a-kind (Hameri, 1997), design-to-order and engineer-to-order (Hicks et al., 2000) are frequently used in combination with supply chain, production, organisation and system when referring to ETO. Bozart and Chapman (1996), Rahim and Baksh (2003), and Wadhwa et al. (2006) argues that ETO companies offer existing designs modified to order, whilst several other authors such as Bertrand and Muntslag (1993), Wortmann (1995), Little et al. (2000), and Vrijhoef and Koskela (2000) argues that ETO companies offer completely new designs developed to order. This suggests that there is a lack of agreement as to what degree ETO companies offer customised products. In spite of this, most authors agree that all production dimensions in an ETO company is customised for each order, that the CODP is located at the design stage, and that the companies operate in project environments with project specific demand (Bertrand & Muntslag, 1993; Hicks et al., 2000; Hicks et al., 2001; Martínez-Olvera & Shunk, 2006; J. Olhager, 2003b; Samadhi & Hoang, 1995; Thompson et al., 1998; Vrijhoef & Koskela, 2000; Wikner & Rudberg, 2005; Wortmann, 1995). This essentially means that there is a consensus on that ETO companies are both dynamic and uncertain, and must therefore be flexible.

2.1.3.1.3.1 Market related characteristics in the ETO environment *Delivery lead-time requirements* 

Generally, high levels of customisation leads to increased costs, higher risks and long leadtimes (Hicks et al., 2000). Konijnendijk (1994) pointed out that one of the biggest coordination problems in an ETO company is setting lead-times. To solve this problem, marketing and production must coordinate both the initial lead-time given to the customer, as well as any change in the product or planning that will influence this lead-time. However, in an ETO environment, it is likely that the customer will be prepared to wait longer in order to receive a fully customised product (Rahim & Baksh, 2003).

#### Demand volatility

Most products produced by ETO companies belong to low volume industries and is therefore exposed to demand volatility, which reduces the opportunities for advanced planning (Hallgren & Olhager, 2006; Rahim & Baksh, 2003). This is further complicated as the ETO environment, compared to other environments, is more prone to demand sensitivity in regard to macro-economic fluctuations (Bertrand & Muntslag, 1993; McGovern et al., 1999). Because an ETO company has a customer order driven production, dealing with this volatility by means of for example creating capacity stock, will run the risk of that stock becoming obsolete (Rahim & Baksh, 2003). Thus, unknown sales and product specifications for future customer orders makes it nearly impossible for an ETO company to conduct forecasts (Hicks & Braiden, 2001).

#### Product range and customisation requirements

Previous orders usually make up the product range of ETO companies. Because orders are more or less customer specific, every order is unique, and ETO companies therefore have a wide product range (Childerhouse & Towill, 2000; Hicks & Braiden, 2001; J. Olhager, 2003b). Product innovation is often led by the need to meet new customer requirements (Hicks et al., 2000). In terms of an ETO product there are often very strict regulatory requirements and design codes that must be fulfilled, especially regarding safety and reliability. But most important are the functional requirements of the product in question (Rahim & Baksh, 2003).

#### Customer order size and frequency

Each customer's order within ETO companies is considered to be large, as each unit of demand represents large portions of the design and production capacity (Hicks & Braiden, 2001). The customer order frequency in an ETO company tends to vary due to the fact that

the company delivers products with unstable demand, which are customised to fit the customer's requirements (Gosling, 2011). Gosling et al. (2013) argues that one of the biggest challenges of an ETO company is uncertainty regarding the customer orders.

# 2.1.3.1.3.2 Product related characteristics in the ETO environment *Customisation opportunities offered*

For an ETO company the customers are its main objective, and the company is dependent on the ability to adapt to customers' requirements (Rahim & Baksh, 2003). The majority of products produced in an ETO company are highly customised and can be seen as completely new designs developed to order. However, modification of previous designs are also common (Gosling & Naim, 2009). The close involvement of the customer often results in frequent engineering changes, and indicates that unpredictability in the production characteristics are common in the ETO environment (Konijnendijk, 1994). Very often a customer will change its requirements along the process, for example a modification in shape or material, which will lead to changes in the product specification (Huang et al., 2001).

#### Material profile and product structure complexity

Driven by customer requirements, products produced in ETO companies are highly complex and they therefore have a multileveled product structure (Grabenstetter & Usher, 2013). A product's structure is reflected in its bill-of-material (BOM), which is a list that contains exact information on all parts or materials required to manufacture one unit of a specific endproduct (Moustakis, 2000). Each different product, made by a specific manufacturer, will have its own separate BOM. For an ETO company the BOM is wide and multileveled, reflecting the diversity of components required and the extensive amount of assembly processes required, which implicates long lead times (J. Olhager, 2003b; Slack et al., 2004, p. 494).

# 2.1.3.1.3.3 Production related characteristics in the ETO environment *Production lead-time*

ETO products long lead times pose a major constraint on the CODP. In an effort to reduce lead-time to a lead-time accepted by the market, ETO companies will in the early stages of a project order components with long lead times. This poses a risk to the ETO company as the complete product structure tends to be uncertain in the early stages of a project and because customer requirements may change throughout the project (Bertrand & Muntslag, 1993; J. Olhager, 2003b).

#### Number of planning points

Processes such as sale, production and engineering, tend to be complex and time consuming for an ETO company (Rahim & Baksh, 2003). In an ETO company it is therefore necessary to coordinate the different processes, and balance the customer's requirements with the production capabilities for each customer order. Because of this, and because ETO companies produces low volumes of highly customised and unique products with long lead times, their production process tend to be structured as projects (Hickson et al., 1969; Silvestro et al., 1992; Slack et al., 2004, pp. 111-116). Such structures are considered to be a continuous process and a single production unit (Bertrand & Muntslag, 1993).

#### Flexibility

To keep up with the developments in the volatile markets, a company's ability to quickly adapt itself and its processes is crucial (Gosling et al., 2013; Hicks & Braiden, 2001; McGovern et al., 1999). This is especially important as the customer requirements within an ETO environment vary from customer to customer and because the different production characteristics will most likely vary from one product to another (Gosling & Naim, 2009). To overcome this challenge there is a strong need for flexibility, which according to Gosling et al. (2013), Little et al. (2000) and Rahim and Baksh (2003) is the answer to quick adaptability as well as an ETO company's key to success (Rahim & Baksh, 2003). McGovern et al. (1999) argued that because an ETO company would be able to understand and adapt customer's requirements, they hold a major competitive advantage.

#### Bottleneck position

Within an ETO company, every order starts in the engineering department, and tend to stay there for a large amount of time. This is due to the fact that designers are reluctant to release the BOM information too early in fear of receiving requests in engineering changes after it is released. The result of this is that the purchasing department often receives the necessary information for doing its job too late and therefore is not able to achieve good prices or to purchase the required items in time. The engineering function is therefore considered to be the main bottleneck within ETO companies (Cutler; Grabenstetter & Usher, 2013; J. Olhager, 2003b; Xu & Liang, 2011).

Olhager's (2003b) determinants of operations strategies also included other characteristics such as seasonality of demand, modular product design and sequence dependent set-up times. However, since these characteristics are not relevant for an ETO company, they are not elaborated in this thesis.

## 2.2 Strategic fit

As mentioned in Section 2.1.3.1.2, a company's operations strategy, and product-, marketand production characteristics determine the company's approach for managing its processes. Achieving a strategic fit between these elements is a key determinant of the company's ability to increase its performance and achieve long-term success (Barnes, 2008, p. 21; Ortega et al., 2012; Wagner et al., 2012).

Chopra and Meindl (2007) indicates that a company must achieve good understanding of three aspects: the customers needs, the degree of demand uncertainty within the environment the company operates in, and the limitations imposed by the characteristics of the company's products. As Chopra and Meindl's (2007) model on strategic fit (cf. Figure 2.4) indicates, when a company has achieved a good understanding of these aspects (horizontal axis), the company can determine how to appropriately respond to demand through its operations strategy in the form of an appropriate degree of responsiveness (vertical axis). By doing this the company can ensure that its operations strategy is within the zone of strategic fit (Chopra & Meindl, 2007, pp. 24-38). Based on product-, market-, and process characteristics products can be characterised as either certain or uncertain, while the operations strategy can be characterised as either efficient or responsive. Only when there exist a strategic consistency between a product's demand supply and demand characteristics, strategic fit is achieved (Wagner et al., 2012). Based on the characteristics of an ETO company identified in Section 2.1.3.1.3, this implies that a company with an ETO approach that offers a large variety of products to its customers, will increase the implied uncertainty because the demand per product becomes more disaggregate. Thus a company with an ETO operations strategy can achieve strategic fit for products with uncertain demand by improving its responsiveness.



Figure 2.4: Finding the zone of strategic fit (Chopra & Meindl, 2007, p. 32).

The strong link between the market requirements and operations strategy concerns the choice of a PPC system. If the wrong PPC system is chosen, a misfit can occur between the market requirements and the operations strategy. This in turn can affect flexibility and cost as well as result in decreased performance and competitiveness, thus underlining the strategic importance of choosing the correct PPC system (W. Berry & Cooper, 1999; J. Olhager, 2003a; Slack et al., 2004, p. 120).

## 2.3 Impact of PPC systems

While market requirements defines a company's performance measures, a PPC system defines the company's ability to accommodate its market requirements (Cheng & Musaphir, 1996). It is therefore vital that there is a good fit between the PPC system and the company's performance measures. The impact of the PPC system on company performance can therefore be assessed considering the five performance measures identified in Section 2.1.2.1 (Gianesi, 1998).

Quality is closely linked to customer satisfaction, and refers to doing things right (Slack et al., 2004, p. 45). Higher quality can reduce the amount of mistakes and thus reduce costs and increase dependability as the stableness and efficiency of a company's processes are improved. The costs will therefore have an external impact on customer satisfaction, while

dependability will have an internal impact (Gunasekaran et al., 2004; Slack et al., 2004, pp. 45-46).

Speed refers to doing things fast, and is concerned with how quickly a producer can respond to demand signals (Slack et al., 2004, p. 47). There are several appropriate ways to measure speed in regards to PPC systems. One of them is delivery lead-time, which is the time it takes from a customer places an order until the customer receives a complete delivery from the supplier. A speedy delivery has an internal impact as it can increase productivity, but also an external impact as customers are willing to pay more to have an order delivered quickly (Ferdows & Meyer, 1990; Gunasekaran et al., 2004; Slack et al., 2004, pp. 47-49).

Dependability refers to doing things on time, such as delivering a customer order when it is promised. It can be reflected by large variations in service levels, meaning that customers cannot always rely on receiving their complete orders on time. Dependability can have both internal and external impacts; such as saving time, cost reduction and providing stability (Ferdows & Meyer, 1990; Slack et al., 2004, pp. 49-50).

Flexibility refers to the ability of changing an operation in any possible way and is crucial for coping with demand uncertainty, responding to different types of variability and to be able to customise orders. For handling demand uncertainty, forecast accuracy is an appropriate measure, whilst safety stock is often used to mitigate stock outs and thus protect against demand variability. Flexibility has internal effects, in which it increases responsiveness, saves time, and maintains dependability (Gunasekaran et al., 2004; Slack et al., 2004, pp. 52-54; G. P. White, 1996).

Cost refers to profitability and the ability of producing at low cost. An improvement in cost can either increase profitability of existing volume of products, or reduce the contract price to sell higher volumes. There is high interdependency between cost and the other measures mentioned, as an improvement in the performance of one of them often lead to a cost improvement for the company as well (Barnes, 2008, p. 25; Gunasekaran et al., 2004; Slack et al., 2004, pp. 54-55). The five identified tangible measures and how to measure their performance in PPC systems are presented in Table 2.1.

In addition to the tangible measures, subjective measures should also be included when

evaluating a PPC system to uncover the systems weaknesses and to get a better understanding of its performance results (Sum et al., 1995; E. M. White et al., 1982). Subjective measures are of interest as White et al. (1982) discovered several frequent problems encountered during PPC system implementation (cf. Table 2.1). Whilst lack of top management support was due to low support from finance, manufacturing, marketing and the head of the departments, the problems with PPC systems evolved around lack of clear goals and company expertise, as well as the lack of communication of these within the company (E. M. White et al., 1982).

Performance measures			
		Delivery promise	
	Quality	Amount of defect products	
		Number of customer complaints	
		Failure cost	
		Frequency of delivery	
		Delivery lead-time	
se	Speed	Production response time	
asur		Physical throughput time	
Tangible me		Inventory turnover rate	
	Dependability	Frequency of late or imperfect deliveries	
		Average lateness of orders	
	Flexibility	Forecast accuracy	
		Level of safety stock	
		Excess capacity expressed by the amount of overtime	
	Cost	Number of full-time workers	
		Cost of products	
		Value added	
	Education of pe	rsonnel	
SS	Lack of top management support		
sure	Implementation approach (Lack of time, personnel)		
mea	Problems with the PPC system (Lack of technical expertise)		
stive	Gaining acceptance		
ıbjec	Inventory control and record		
Su	Accuracy		
	Forecasting demand		

**Table 2.1:** The tangible and subjective performance measures. The table is adapted from Slack et al. (2004, p. 643).

This framework can be used to evaluate a PPC systems performance and provide a better understanding of its performance results. The framework will therefore be used as a tool for structuring and analysing the quantitative and qualitative data in Chapter 6.

## 2.4 Material requirements planning

This section provides a review of another major topic within this research, the PPC system material requirements planning (MRP), to help answer the research topic of the master thesis.

#### 2.4.1 Material requirements planning

An important function of a PPC system is MRP. MRP is a computer-based planning and control system whose main objectives are to provide the right part at the right time, and to meet the schedules for completed products (Hill & Hill, 2009, p. 324; Vollmann et al., 2005, p. 222; E. M. White et al., 1982).

#### 2.4.2 Evolution of MRP

By the 1960s, the main competitive focus within the manufacturing industry was price. Inventory management systems such as reorder point and economic order quantity (EOQ) was predominant within the manufacturing industry (Islam et al., 2013; Morabito et al., 2005). Although these systems addresses the amount of inventory that must be purchased or produced as well as determines the reorder point, they do not address the "when needed" aspect of the inventory (Mabert, 2007, p. 346). This led to the development of MRP.

Joseph Orlicky is considered to be the "father" of MRP (Mabert, 2007, p. 348). In 1964 Orlicky developed MRP after studying TOYOTA's manufacturing program. This revolutionised the manufacturing industry by enabling companies to plan for needed material based on an overall master schedule through a BOM. The same year as MRP was developed, Black & Decker became the first company to use the system, and since then MRP has become widespread throughout the manufacturing industries (Islam et al., 2013). The development of MRP has evolved into the more sophisticated and complex manufacturing resource planning (MRP II) and enterprise resource planning (ERP). However, MRP still remain the core of these systems (Ptak & Smith, 2008). Today, MRP is the most widely used PPC system in the world (Sum & Ng, 2014).

#### 2.4.3 Basic concepts of MRP

The aim of MRP is to ensure fulfilment of customer orders by releasing a set of production orders for each item of the BOM, which allows synchronizing the internal and external logistics flow. For this purpose, the stock at hand, the planned receipts and the gross requirements determined from the master production schedule (MPS), which is the driving

force of an MRP system, provides the basic calculations computed for each item of a future inventory level forecast (D'Avino et al., 2013). However, the most important function of the MRP system is considered to be its ability to reschedule when unplanned events occur. When updated, this rescheduling function attempts to maintain the alignment between the order due date and the order need date (C. J. Ho, 1989, 2002; C. J. Ho et al., 1995).

A diagram of a basic MRP system can be seen in Figure 2.5. The figure shows that an MRP system has three major input components: the MPS and the BOM, as already mentioned, but also inventory data. Inventory data keeps an up-to-date record of each item in the inventory and contains information such as quantity at hand, safety stock level, and procurement lead-time. The inventory data is updated whenever an inventory transaction is made (Avraham & Karni, 2013; Roy, 2005, p. 141). Based on these MRP inputs the MRP system is able to determine what work orders and purchase orders that are needed and when these orders must be placed to ensure that all materials arrive when needed.



Figure 2.5: The basic MRP system. The figure is adapted from Avraham & Karni (2013).

#### 2.4.4 Prerequisites and assumptions of MRP

According to Roy (2005) and Bertrand and Muntslag (1993) there are several prerequisites and assumptions that need to be fulfilled for an MRP system to function optimally.

- A MPS must exist and be stated in the BOM. Thus allowing companies to combine planned parts and forecast for any level of the BOM.
- The system requires that a BOM must exist at planning time, and also assumes that the BOM is both completed and accurate for when it is released into the system.
- Individual item lead-times are known and can be supplied to the system, at least as estimates. These estimates are used as a basis for the MRP system to provide the right part at the right time.
- All inventory items are required to have their unique part number, as the MRP system assumes that identical part numbers are interchangeable.
- For an assembly to be released for production, all components of that assembly must be available at the time of assembly order release.
- MRP assumes processes independence between items. This means that the manufacturing of any given inventory item can be started and completed on its own and is not dependent on other orders for completing the process.
- The system requires inventory records containing data on the status of every available item, and assumes that all items goes in and out of stock.
- All processes must be of physical nature.

The most important requirement of the MRP system however, is that file data inserted to the MRP system is accurate, complete and up to date as it facilitates full traceability back to the BOM of the project (Roy, 2005, p. 138).

### 2.4.5 MRP advantages

When an MRP system is successfully implemented there are several benefits that can occur. Because an MRP system provides accurate control over the many complex activities that a company perform, these activities are synchronised so as to ensure that parts are manufactured when they are needed in the next stage of production. By doing this, the system ensures no shortages of components in inventory and eliminates uneven utilisation of resources (Roy, 2005, p. 152).

Also, several studies that have been conducted on the usage of MRP systems have found the following tangible advantages to be the most common: increased inventory turnover, better delivery lead-time, increased percentage of time meeting delivery promises, reduced percentage of orders requiring "splits" because of unavailable materials, and reduced number

of expeditors (Aghazadeh, 2003; J. C. Anderson et al., 1984; Petroni & Rizzi, 2001). Subjective advantages that often occur are improved competitive position, improved product quality, improved productivity, increased information on which to base decisions, better production scheduling, reduced safety stock, better cost estimation, and increased accuracy of BOM, MPS and inventory (Sum & Yang, 1993; Sum et al., 1995).

#### 2.4.6 MRP drawbacks

Although MRP systems are widely used within manufacturing companies and has several benefits, they also exhibit significant drawbacks (D'Avino et al., 2013). The MRP system is largely constrained by the product structure, lot sizing and frequency of updates (Roy, 2005, p. 134).

One of the drawbacks is that the MRP system uses deterministic data to develop plans (Hicks & Braiden, 2001). It therefore ignores real life uncertainties that can occur and affect the production. These uncertainties can for example be shortages and variability in materials and labour, faulty components, and uncertainties in respect to lead-times of materials. When these uncertainties occur, it may affect cost and customer delivery performance (Koh et al., 2000).

However, one of the most significant drawbacks of the MRP system is "nervousness". Nervousness relates to the rescheduling function of the MRP system (D'Avino et al., 2013). MRP system nervousness is described in several ways in the literature (Blackburn et al., 1986; Carlson et al., 1979; Minifie & Davis, 1990; Penlesky et al., 1989). What these descriptions all have in common is that MRP system nervousness can be seen as instability in or excessive rescheduling of planned and open orders (C. Y. Ho, 2007).

According to Blackburn et al. (1986), Lee and Adams (1986), Wemmerlöv (1989), and Ho and Ireland (1993), MRP system nervousness is caused by various variables. Environmental variables, such as demand variability, capacity utilisation and cost structure, are factors beyond the operating system, and thus decision makers have little or no control over them. Whilst operating variables, such as lot sizing techniques, length of lead-time and length of planning horizon are controllable and exist within the operation system. Any uncertainty in the production systems can cause planned- or open-orders to be rescheduled in the MRP system, so as to be able to maintain up-to-date priorities. However, too frequent rescheduling changes can result in excessive rescheduling and outweigh the benefit of maintaining the up-

to-date priority (C. J. Ho, 1989). Large inefficiencies such as excessive inventory, risk of a decrease in customer service, risk of lost revenues, misguided capacity plans and risk of missed production schedules are among the consequences of MRP system nervousness (D'Avino et al., 2013).

There exist several methods for dampening system nervousness (C. J. Ho et al., 1995; C. Y. Ho, 2007; Minifie & Davis, 1990). In an MRP system, production uncertainty can generally be resolved by two approaches; the use of slack or buffer, and the use of rescheduling capability. Conventionally, safety stock, safety lead-time and safety capacity are the most common inventory oriented methods used as slack to buffer the production system, and thus dealing with quantity uncertainty. However, the general weakness of these dampening methods is that the buffer, which can take the form of unused capacity or extra inventory, can be costly to maintain. The alternative is to use the rescheduling capability of the MRP system, and thus realign the due date with the need date (Blackburn et al., 1986; Carlson et al., 1979). This can be done by implementing a lot sizing technique, which cope with rescheduling problems by determining timing and size of order quantities. Most of the techniques require extra inventories that can be used to accommodate schedule changes, and these built-in stocks can resolve small deviations in production schedules without changing the due dates of open orders, and thus reduce MRP system nervousness. However, the cost of carrying these extra inventories must be evaluated against the nervousness dampening effect (C. J. Ho, 2002; C. J. Ho & Ho, 1999; C. Y. Ho, 2007). The choice of a lot sizing technique is therefore a major factor affecting the MRP system nervousness, as the different lot sizing rules determines the amount of excess inventory (Y. Y. Lee et al., 1990). The simplest and most common is the lot-for-lot (LFL) technique. This technique minimizes the holding cost by ordering as little as possible each distinctive period (C. J. Ho & Ho, 1999). However, with no excess inventory, minor changes in plans will lead to MRP system nervousness (T. S. Lee & Adam Jr, 1986; Narasimhan & Melnyk, 1984; Wemmerlöw & Whybark, 1984). Other popular lot sizing techniques are part period balancing (PPB) and economic order quantity (EOQ). These techniques require built-in inventories and therefore facilities dampening of minor schedule changes (C. J. Ho, 2002; Narasimhan & Melnyk, 1984; Ram et al., 2006). Another lot sizing technique is the fixed order quantity technique (FOQ). This technique orders a predetermined quantity whenever an order is placed, and is one of the most common lot sizing techniques used in practice (C. J. Ho & Ho, 1999; Ram et al., 2006). Choosing the correct lot sizing technique is all about balancing the number of orders against the cost of carrying inventory. For instance, LFL will keep average inventory levels low, but will result in more orders on average. While FOQ will have higher inventory levels, but will result in less orders on average. This means that as long as a company has the correct cost information, it is possible to derive the most effective and least costly lot sizing technique to use for any particular item (Narasimhan & Melnyk, 1984).

#### 2.4.7 Factors affecting the successful implementation of MRP systems

Through a number of studies, several researchers have found that there are numerous factors that affect a successful implementation of MRP systems. In their studies, Burns et al. (1991), Kim and Lee (1993), and Samaranayake et al. (2002) all found that organisational factors, which among others include company size, company complexity and operations strategy, must be considered when implementing an MRP system. In addition to these factors, Sum et al. (1995) and Koh et al. (2000) also found that implementation is affected by the degree of uncertainty in the market place. It is also generally believed that MRP implementation is influenced by factors such as years of implementation, degree of data accuracy and extent of problems occurring during implementation (Aghazadeh, 2003; Ang et al., 1995; Sum & Yang, 1993; Sum et al., 1995; E. M. White et al., 1982; Wong & Kleiner, 2001). In addition to these factors some researchers believe that technological factors, such as degree of integration among MRP modules and MRP system features will affect the implementation (Keung et al., 2001; Sum & Yang, 1993). However, the majority of conducted studies points out that MRP implementation problems, more often than not, relates to people and the fact that they are not of technical nature (Burns et al., 1991; Callerman & Heyl, 1986; Jp, 1998; Sum et al., 1995; E. M. White et al., 1982). For example autonomous and highly educated people find it difficult to fit into the rigid MRP structure as they often have a need to unfold themselves in terms of innovation (Tang & Y.C., 2010).

# 2.4.8 Literature review on the use of an MRP system in an ETO environment

There is a lack of literature on the use of an MRP system in an ETO environment. However, this research has taken the same assumptions as Stevenson, Hendry, Kingsman, and Aslan did in their papers (2012; 2005). Although their papers were based on MTO companies, they defined MTO in a broad sense by using it as an umbrella term, thus referring to companies that produce customised products to fit the customer's specifications but that do not repeat the production of the product on a regular basis or in a predictable manner. This means that their

definition of an MTO company included the basic characteristics of an MTO strategy, where the design is already available before an order is received and where the remaining work is manufacturing and assembly. In addition to this, their definition also includes product design, which is applicable for an ETO company. Thus, in their research the MTO term refers to products where production and design do not take place until after the customer order has been confirmed. This means that their MTO term includes ETO, but excludes MTS and ATO, and this assumption will thus be used as a foundation in this research.

As previously mentioned, MRP is the core within ERP. Because of this, this research assumes that some parts of the same literature's view on ERP systems can be used to describe MRP systems. Especially, since literature has found that ERP and MRP share similar implementation problems to achieve a successful implementation (Muscatello et al., 2003).

Several suppliers of MRP systems states that benefits of such a system can be accumulated by any organisation, independent of what type of product or service they may offer, as their systems are configurable to meet the needs of all types of operations strategies. However, literature suggests that a company that produce customised products, such as an ETO company, will possibly encounter challenges when implementing an MRP system (Bertrand & Muntslag, 1993; Deep et al., 2004; Stevenson et al., 2005). Thus, it remains unclear whether MRP can sufficiently accommodate an ETO company's needs. A summary of key studies that partially explore MRP adoption in an ETO context can be found in Table 2.2.

Author	Topic	Operations	Research	Findings
		strategy of	size	
		the firm(s)		
Bendoly and	MRP	Various	N/A <sup>1</sup>	The authors conducted an investigation into the importance of a system being
Jacobs (2006)				aligned with the company's specific process requirements and found that
				overall performance and satisfaction of a company becomes weaker if the
				company's operations strategy is misaligned with the adopted MRP strategy.
Bertrand and	MRP	ETO	N/A	Based on the knowledge that many ETO companies have tried to implement
Muntslag (1993)				MRP systems and ended up with little or no success, the authors conducted a
				conceptual study on the assessment of MRP suitability into the ETO
				environment. Their conclusion was that system functionality is the reason for
				the unsuccessful implementations of an MRP system. The authors also argued
				that the changeable environment and variation in design and production in ETO
				environments might severely affect the applicability of an MRP system in an
				ETO company.
Petroni (2002)	MRP	Various	109	The author conducted a survey and found that the main reasons for MRP
				implementation problems are data accuracy, level of functionality integration
				and level of management support.
Little et al.	MRP	ETO	100	The authors conducted a case study and found that the characteristics of the
(2000)				ETO operations strategy differ substantially from the basic assumptions of an

<sup>&</sup>lt;sup>1</sup> Not available (literature review only)
				MRP system.		
Stevenson et al.	PPC	МТО	N/A	The authors reviewed and assessed classical approaches to PPC and their		
(2005)				applicability to MTO production. Their conclusion was that MRP does not		
				provide sufficient support for managing customer enquiries in a MTO		
				environment.		
Muscatello et al.	ERP	Various	4	The authors conducted a thorough case study and concluded that MRP might		
(2003)				be suitable for small or medium sized enterprises (SME) as MRP suppliers are		
				turning their marketing sights towards these company sizes.		
Mabert et al.	ERP	Various	18	The authors conducted a case study and survey of the impact of organisation		
(2003)				size on ERP adoption in North American companies and found that large firms		
				employ more of the functionalities offered by the system and customise the		
				software more than smaller firms.		
Laukkanen et al.	ERP	Various	44	The authors conducted a study of Finish companies and found that large		
(2007)				companies expect more from the ERP system than small companies.		
Buonanno et al.	ERP	Various	366	The authors conducted a survey of factors influencing ERP adoption in SMEs		
(2005)				compared to large companies. In their research of business complexities,		
				organisational changes and MRP adoption the authors found that company size		
				was the only significant factor affecting ERP adoption.		
Olsen and Sætre	ERP	ETO/MTO	N/A	The authors conducted case studies and found that an ETO company that was		
(2007)				experiencing problems with determining realistic due dates, coping with		
				increasing demand and accommodating the customisations requirements of		

				each order was considering a number of ERP systems but was unable to find a			
				system that was suitable for their problems. The company therefore developed			
				its own in-house design and engineer solution to cope with product			
				customisation and the design and engineer stage, as this was where the biggest			
				problem lay.			
Morabito et al.	ERP	Various	150	The authors conducted a survey of the impact of organisation size on ERP			
(2005)				adoption in Italian companies and found that large firms employ more of the			
				functionalities offered by the system and customise the software more than			
				smaller firms.			
Aslan et al.	ERP	МТО	N/A	The authors conducted a systematic literature review of the fit between ERP			
(2012)				functionality and a MTO operations strategy and found that although ERP can			
				provide benefits to MTO companies, it is obvious that there is a misalignment			
				in important areas. The authors found several key areas that are in need of			
				further research in order to improve this alignment, and especially focus on the			
				link between the decision support provided by ERP systems and the decision			
				support required by MTO companies at both the customer enquiry stage and			
				the design and engineering stage.			
Sriram et al.	Various	ETO	N/A	The authors conducted a conceptual discussion on the key challenges and			
(2013)				opportunities that different PPC systems offer in an ETO environment, and			
				found that MRP lack the capability to meet the requirements of an ETO			
				company.			

**Table 2.2:** Summary of key MRP studies of relevance to an ETO context.

A common factor mentioned in the majority of the above reviewed studies is that there is a lack of literature on the MRP and ETO relationship, and the little literature that has touched upon this relationship has generally concluded that an MRP system will not function optimally in an ETO company. And as Bertrand and Muntslag (1993) pointed out in their study, the most common decision to use an MRP system is that there is a wide availability of MRP software, as well as the fact that it is rarely understood that the MRP software is not suitable for an ETO company. In fact, as the authors briefly explain, the MRP concept was developed for the MTS and ATO strategy. These strategies differ from ETO, as well as MTO, in that the customer plays a central role in the latter two strategies.

# **3 RESEARCH OBJECTIVE AND RESEARCH QUESTIONS**

The literature review conducted in Section 2.4.8 clearly states that there is a need for research on the MRP and ETO compatibility as this will generate knowledge on the applicability of an MRP system in an ETO company. This chapter therefore presents the overall research objective of the thesis (cf. Section 3.1), as well as its corresponding research questions (cf. Section 3.2.).

# 3.1 Research objective

The objective of this research is to investigate if implementation of an MRP system supports the operations strategy of an ETO company. For this purpose, the authors of this thesis have worked with a specific case, NOVN, which is going to implement an MRP system in the near future. As requested by NOVN, part of this research will be conducted on the basis of one unit within the roughneck department delivering roughneck machines. This unit has in the past had an ATO operations strategy, as well as implemented an ATO based PPC system in connection with the commencement of series production. Although this system is not directly comparable to an MRP system, NOVN considers the performance of the ATO based PPC system highly relevant for evaluating whether an MRP system supports their ETO operations strategy. The ATO based PPC system was in use for three years before it was terminated.

The overall research objective of this thesis is to:

Investigate if implementation of an MRP system supports the operations strategy of an ETO company.

# 3.2 Research questions

Research questions help guide and centre the research of the overall research objective (Zikmund et al., 2013, p. 120) To address the research objective of this thesis, the overall research objective is broken down into two specific research questions:

Research question 1 (RQ1): What is the impact of an ATO based PPC system in an ETO company?

This research question is explicitly addressed in Chapter 6. To answer RQ1 research will be conducted on the basis of quantitative and qualitative research of the selected unit within the roughneck department. By analysing the performance of the roughneck unit during the time the ATO based PPC system was in use, it is possible to identify the strengths and weaknesses of the system and identify the impact it had on the unit.

In accordance with the literature reviewed in Chapter 2, a strategic fit between the product-, market- and production characteristics and the PPC system, is positively related to increased performance of a company. Based on this knowledge there is reason to expect that the roughneck units ATO operations strategy had a positive impact on the roughneck units performance, assuming that the ATO based PPC system was supporting the roughneck units ATO operations strategy. To answer this, a quantitative analysis of testing whether the ATO based PPC system has had a positive impact on the roughneck unit has been conducted. The ATO based PPC system serves as the independent variable, while the five tangible performance measures identified in Section 2.1.2.1 and elaborated on in Section 2.3: quality, speed, dependability, flexibility and cost serves as the dependent variables. Within NOVN, value added is considered to be an important performance measure. In this research, the value added is therefore treated as a separate measure constituting the sixth dependent variable. The list of measures applied in this analysis has been adapted in accordance with the available data retrievable from NOVN and computational resources. These dependent variables are discussed, the independent variables expected influence on NOVN explained, and related hypotheses are drawn.

## Quality

As explained in Section 2.3, quality is closely linked to customer satisfaction and depends on how well the product fits patterns of customer preferences. For evaluating what impacts the ATO based PPC system has had on product quality, internal failure costs have been identified as an appropriate measure. Internal failure cost can be defined as the cost of poor quality caught before the product leaves the production facility and that can lead to rework, scrap and lost production hours. Reductions in internal failure cost can therefore enhance profitability (K. Boyer & Verma, 2010, p. 30). Within NOVN unforeseen extra working hours is a common measure of failure cost. The implementation of the ATO based PPC system was expected to enhance product quality, as the system would free up resources that could be redirected in order to enhance product quality. The ATO based PPC system should therefore have a significant impact on the product quality of the roughneck unit.

### The following hypothesis is formulated:

*H*<sub>1</sub>: *The ATO based PPC system led to a reduction in unforeseen extra working hours.* 

### Speed

The delivery lead-time, the time it takes from a customer places an order until the customer receives a complete delivery from the supplier, has been identified as an appropriate measure to investigate if the ATO based PPC system has had an impact on the ability of the roughneck unit to quickly respond to demand signals. As the roughneck units decoupling point was shifted forward from ETO to ATO, the time to market was expected to be reduced. The implementation of the ATO based PPC system was therefore expected to reduce delivery lead-time for roughnecks.

### The following hypothesis is formulated:

*H*<sub>2</sub>: *The ATO based PPC system led to a reduction in delivery lead-time.* 

## Dependability

While speed refers to the lead-time of a product, dependability refers to the ability to deliver the product in time of the contractual delivery date. Dependability can be reflected by large variations in service levels, meaning that customers cannot always rely on receiving their complete orders on time. Dependability can have both internal and external impacts; such as saving time, cost reduction, providing stability and developing fruitful customer relations. Frequency of late or imperfect deliveries is one of the identified measures that will reflect if the ATO based PPC system has had an impact on the roughneck unit's delivery reliability. NOVN regards late deliveries as imperfect deliveries, as they result in penalty costs. The dependability of the roughneck unit was expected to be enhanced by the ATO based PPC system as the new system enabled the unit to allocate components scheduled for a specific project to other projects depending on the project urgency.

### The following hypothesis is formulated:

H<sub>3</sub>: The ATO based PPC system led to reductions in frequency of imperfect deliveries.

### Flexibility

Flexibility refers to the ability to respond to changes in the market. The ATO based PPC system was expected to increase the roughneck unit's flexibility and enable it to respond to changes in for instance a delay in delivery lead-time. Operational measures of flexibility are accuracy of forecast, level of safety stock and capacity utilisation. As the roughneck unit has an ETO operations strategy, no forecasts are conducted and raw material and components are not ordered before the order has been placed by the customer. Accuracy of forecast and level of safety stock will therefore not be a good measure in this setting. Additionally, the registered data on overtime hours on the roughneck projects are inadequate, making also this measure problematic to use. On the other hand, hours elapsed per roughneck project can provide an indication of the flexibility of operations within the roughneck unit, as a decrease in hours could indicate excess capacity.

### The following hypothesis is formulated:

*H*<sub>4</sub>: *The ATO based PPC system led to a decrease in working hours per roughneck.* 

### Cost

Cost refers to profitability and the ability of producing at low cost. A reduction in cost can either increase profitability of existing volume of products, or reduce the contract price to sell higher volumes. The ATO based PPC system was expected to improve the efficiency and optimize the use of available capacity and hence reduce cost per roughneck.

### The following hypothesis is formulated:

*H*<sub>5</sub>: *The ATO based PPC system led to a reduction in costs per roughneck.* 

### Value added

Value added refers to the ability to attribute a rather homogeneous product a greater sense of value perceived by the customer without necessarily increasing the cost of the product (Doyle, 2011). Such value added can lead to an increase in contract price. It is therefore a measure of profitability and is calculated as the profit margin. The ATO based PPC system was expected to reduce delivery lead-time for roughnecks, and thus add value for NOVN's customers. As NOVN considers delivery lead-time to be one of the main order qualifiers, potential customers of NOVN was anticipated to perceive shorter lead-times as value adding which would result in higher contract prices and thus higher profit margins.

The following hypothesis is formulated:

*H*<sub>6</sub>: *The ATO based PPC system led to an increase in profit margin.* 

This leads to the following research model (cf. Figure 3.1):



Figure 3.1: Research model.

As explained in Section 2.3, to get a better understanding of the ATO based PPC system's performance, subjective measures should be included when evaluating the PPC system. Qualitative research based on these subjective measures will be conducted on the basis of the roughneck unit's employees.

Research question 2 (RQ2): *How can an MRP system be used to support the PPC needs of an ETO company?* 

This research question is explicitly addressed in Chapter 7. The roughneck unit's employees and other key informants within NOVN will be the subject of this research question. An analysis of the unit's employees experience with the system, as well as key informants that are thought to be knowledgeable and credible on the research at hand will unveil the main reasons for the system's performance and help identify potential challenges that can arise during implementation of the new MRP system. This will provide the foundation to explore how an MRP system can be used to support the needs of NOVN.

# **4 RESEARCH METHODOLOGY**

This chapter presents and elaborates on the research methodology used in the thesis for answering the research questions developed in Chapter 3. The research questions are revisited and the nature of investigation for each of the research questions is reviewed.

Section 4.1 introduces the research design used in this thesis, as well as their relevance to the two research questions. Section 4.2 critically reviews and justifies the research methods used in the research. Section 4.3 gives a short description of the sampling unit of the research. Section 4.4 explains how the data was collected, as well as what data that was used to answer the research questions. Section 4.5 provides an assessment of how the research will achieve good scientific validity and reliability. Section 4.6 concludes the chapter with a table that link the research questions with research methodology.

# 4.1 Research design

The research design constitutes the master plan that specifies the methods and procedures for collecting and analysing the necessary information to achieve the objective of a study. (Zikmund et al., 2013, p. 64). There are three types of business research designs: exploratory, causal and descriptive. The three types have different approaches, and it is therefore important that the choice of research design matches the characteristics of the research problem, to obtain useful research results (Zikmund et al., 2013, p. 52). Based on what data was needed to answer the research questions of this thesis, the research conducted has been a combination of exploratory and descriptive research.

Exploratory research is often used to clarify vague situations or to discover ideas that may be potential business opportunities. It can also be used to help identify the research problem. This type of research does not have the purpose to provide conclusive evidence from which to determine a particular course of action, but provides more knowledge and better understanding of the research problem. It can also lay the foundation for further research (Zikmund et al., 2013, p. 52).

Descriptive research is used to describe characteristics of objects, people, groups, organisations or environments. In contrast to exploratory research, descriptive research is

undertaken after the variables in question have been clarified and understood. A good descriptive research depends on stating clear hypotheses on how the variables in question interact with each other (Zikmund et al., 2013, p. 53).

The purpose of RQ1 is to explore what is the impact of an ATO based PPC system in an ETO company. The phenomenon of performance and how to measure performance of a PPC system is well explored in the literature (cf. Section 2.3). A descriptive approach has therefore been applied to address this research question. An assessment of relevant available literature on the phenomenon has been conducted. The consolidation of this literature makes up the base of the performance measures used to evaluate the ATO based PPC system's performance in the unit in question.

The purpose of RQ2 is to explore how an MRP system can be utilised to support the PPC needs of an ETO company. To get a comprehensive understanding of how PPC is managed in NOVN today and to identify challenges of the current PPC system, an exploratory approach has been applied. Furthermore, this knowledge is also the source of suggestions for additional elements that should be considered when implementing the MRP system in NOVN.

The overall focus of the thesis is to "investigate if implementation of an MRP system supports the operations strategy of an ETO company". The nature of this research topic suggests that the research is a combination of deductive and inductive reasoning, as is reflected by RQ1 and RQ2 (Saunders et al., 2009, p. 127). While deductive research is associated with deriving a conclusion about a specific instance based on a known general premise, inductive research is about establishing a general proposition on the basis of observations of particular facts (Zikmund et al., 2013, pp. 43-44). A variety of literature exists on benefits of a PPC system and how to measure it, which supports a deductive reasoning for addressing RQ1. However, because there is a lack of literature regarding the fit between an ETO company and an MRP system (cf. 2.4.8), RQ2 will be addressed by using inductive reasoning. Thus, given the descriptive and exploratory research designs and the nature of the research topic, this thesis is a mixed method research. The chosen research methods will be elaborated in the next section.

# 4.2 Research methods

Both quantitative and qualitative research methods have been chosen, as they are seen as the most appropriate way of conducting the research. Necessary data has been gathered by

conducting a literature review and a case study to provide the foundation for answering the research questions.

## 4.2.1 Literature review

A literature review is a summary of previous research that is relevant to the topic at hand (Zikmund et al., 2013, p. 63). To get a better understanding of the phenomenon of this research, a comprehensive literature review has been conducted. It provides a historical background of the phenomenon as well as identifies knowledge gaps, degree of agreement on topic and the current status of research on the topic.

## 4.2.2 Case study

A case study is an empirical inquiry that investigates a contemporary phenomenon within its real-life context, especially when the boundaries between phenomenon and context are not clear (Yin, 2012, p. 7). It is often used to explore existing theory (Saunders et al., 2009, p. 147). A case study approach has therefore been applied to address the research objective of the thesis. The starting point for the case selection was an opportunity to collaborate with NOVN, who found the topic highly relevant and interesting. As NOVN is the only company studied, this research is classified as a single case study. The benefit of conducting a single case study is that it is possible to delve into the organisation. However, it may be problematic to generalise the findings of the research in this thesis, both statistically and analytically, as it only represent a single organisation and event (Bhattacharyya, 2006, p. 72; Yin, 2009, p. 38).

The case study has been conducted doing both quantitative and qualitative analysis. A quantitative quasi experiment has been applied to provide insight into the operational aspect of the research questions, while personal- and group interviews have been conducted to provide rich descriptions and insightful explanations of the research problem. In addition to the interviews, several informal conversations with company representatives have been carried out.

### Quasi experiment

A quasi experiment aims to evaluate interventions and demonstrate causality of the interventions and their outcome (Cook & Campbell, 1979, p. 95). To describe the impact of the ATO based PPC system on the production of roughnecks, a quasi experiment with one group pre-test and post-test design, using a double post-test, has been conducted. Adding a

second post-test facilitates ruling out maturity effects that may have explanatory power on the observed result (Cook & Campbell, 1979, p. 95). As the ATO based PPC system is no longer in use, it was not possible to make direct observations before or during its intervention. Therefore, archival records with performance data retrieved from NOVN's databases for the three periods constitute the data included in the experiment. This data has been further analysed through the statistical analysis tool, SPSS. An ANOVA have been used to test if there exist significant variances between the three measured periods (Zikmund et al., 2013, p. 542). Scheffe's test have been used to further test the significance of the means to establish between which periods of time there exist significant changes (Klugh, 1986, p. 287). A confidence interval of 95% is considered to be satisfactory to support or reject a hypothesis.

The main weakness of a quasi experiment is its lack of randomisation as it is difficult to control for variables that might have an impact on the results. Variables that may have an impact on the results of this experiment are productivity, demand, quality of employees, as well as the financial crisis. As quasi experiments tend not to be sufficient for permitting strong tests of causal hypotheses, the aim of this quasi experiment was to merely provide support for the causal inference (Cook & Campbell, 1979, p. 95).

### Interviews

The interview is one of the most important sources of case study information (Yin, 2009, p. 106). A personal interview allows the informants to develop and give reasons for the choices that were made in a non-intimidating way (Saunders et al., 2009, p. 372). For the purpose of getting a thorough understanding of the case study issues and the key informants' perspectives on the subject, three personal interviews were conducted. This method was also selected to get an insight into the key informants opinions on the sensitive topic of the reason for why the ATO based PPC system was shut down.

As pointed out by Zikmund et al. (2013, p. 142), an interview with multiple key informants, as opposed to a personal interview, allows for higher degree of scrutiny and detailed descriptions of a given research. It may also produce thoughts around the subject that would not otherwise be produced. To get multiple perspectives on the subject of the research, and for identifying similar and rival opinions, the authors of this master thesis chose to follow the advice of Zikmund et al. (2013, p. 142) and conduct a group interview. Although it is preferable for such an interview to consist of 6-8 key informants, it was only possible to

obtain two. The reason for this was that there were only a few employees remaining in NOVN that had central roles during series production in the given unit.

The main concern of conducting a group interview with only two key informants is that it may be difficult to discuss sensitive topics. On the other hand, a group interview can make key informants more willing to share their thoughts on sensitive topics as they become aware of other informants in the group with similar concerns and opinions as themselves (King & Horrocks, 2010, p. 63). Another concern is that the key informants answers may be coloured by the answers or the presence of other informants and therefore might not reflect their true answer (King & Horrocks, 2010, p. 62). To mitigate these concerns, anonymity was offered.

The conducted interviews of the five key informants, both the three personal interviews and the group interviews, have also shaped further questions that have been found relevant to the topic and appropriate to explore further.

# 4.3 Sampling

The sampling units consist of five key informants. Three of them were employed in the unit in question during the use of the ATO based PPC system, whilst the remaining two has extensive knowledge on series production and MRP. These are all informants that are thought to be knowledgeable and credible on the research at hand.

# 4.4 Data collection and analysis

# 4.4.1 Data collection

Multiple sources of evidence have formed the basis of this research. Articles published in reliable scientific journals and documents including annual reports, company presentations, and other internal reports provided by NOVN, were a rich source of evidence and thus constitute the theoretical data of this research. Qualitative and quantitative data has mainly been collected from NOVN's information system, and through conducting interviews with the key informants. The quantitative data necessary for analysing the performance of the ATO based PPC system was extracted from NOVN's information system. The interviews were recorded with the consent of the key informants to facilitate documentation. This allowed the researchers to fully focus on the answers the key informants provided. In addition, notes were

taken during the interviews. This qualitative data was backed up by informal conversations and interactions with company representatives and other informants.

## 4.4.2 Data analysis

Data analysis consists of making sense of the data collected by converting quantitative and qualitative data into a format that can be further analysed to draw conclusions (Yin, 2009, p. 126). The general analytical strategies for this research rely on theoretical propositions, using both qualitative and quantitative data and examining rival explanations.

## RQ1: Literature review and case study (quasi experiment, personal- and group interview)

The first research question focus on to what is the impact of the ATO based PPC system in an ETO company. A literature review was conducted to identify operational measures and develop a theoretical framework for how to measure performance of a PPC system. In addition, the main reasons for success and failures of PPC implementation were identified to describe the performance results of the ATO based PPC system. By using keywords and backwards literature search, relevant articles on the topic from reliable scientific journals were identified.

Necessary quantitative data has been extracted from NOVN's information system and summarised statistically before they have been analysed through conducting two different F-tests: ANOVA and Scheffe's test. An F-test determines whether there is more variability in the scores of one sample, than in the scores of another sample. It assumes that the means of a given set of normally distributed observations, having the same standard deviation, are equal (Zikmund et al., 2013, p. 546).

## ANOVA

ANOVA tests if there exist significant variances between populations with two or more treatments (Zikmund et al., 2013, p. 542). To test whether there are significant variances between the three periods of time used in this research, an ANOVA test has been conducted. The output from SPSS provides the test with its given F-value as well as calculates the critical value for each test, which is the value the F-test must exceed to reject the null hypothesis (Zikmund et al., 2013, p. 546). The significance of the result is reflected in the p-value. For this study a confidence interval of 95% is considered to be satisfactory to support or reject a

hypothesis. The null hypothesis should therefore be rejected if the p-value of a test is below 0.05.

### Scheffe's test

While the ANOVA test is used to examine if there exist a difference between more than two groups, Scheffe's test examines which groups are significantly different (Klugh, 1986, p. 287). Scheffe's test have therefore been used to further test the significance of the means in order to establish between which periods of time there exist significant changes. Similar to the ANOVA test, SPSS calculates the given F-value and calculates the critical value in which the F-value must exceed to reject the null hypothesis; however, this is not displayed in the output from SPSS. However, the p-value reflects the significance of the test. Furthermore, a confidence interval of 95% is considered to be satisfactory to support or reject a hypothesis. The null hypothesis is therefore to be rejected if the p-value of a test is below 0.05.

The personal- and group interviews provide insights into more subjective measures of the PPC implementation. A content analysis of this data was conducted to identify corroborations and rival explanations, and to draw correct conclusions.

### *RQ2: Literature review and case study (personal- and group interviews)*

The second research question focus on how the MRP system can be used to fit the PPC needs of an ETO company. Insights from the case study, in the form of personal- and group interviews, were used to map the characteristics of NOVN's need for PPC. Based on these characteristics, a literature review has been conducted to identify how an MRP system can accommodate these needs, as well as the potential challenges with implementing this system. These findings are then thoroughly discussed before conducting logical reasoning and recommendations.

# 4.5 Quality of the research design

The quality of this research design is assessed on the following criterions: external validity, internal validity, construct validity, and reliability (Yin, 2009, p. 40).

## 4.5.1 External validity

Generalizability is one of the main concerns when conducting a case study. In some cases, it is possible to generalise statistically to a population based on multiple case studies (Yin, 2009,

p. 44). As this research is a single case study it is not considered to be a representative sample regarding statistical generalization. The purpose of this research is therefore not to produce a theory that can be generalised to all populations, but merely describe what is happening in this particular research setting. (Saunders et al., 2009, p. 158). However, parts of this research may be of an analytical generalizable character if the findings of the research are supported by two or more case studies. In such situations replication may be claimed and hence be analytically generalizable (Yin, 2009, p. 38). Furthermore, the fact that a traditional MRP system is used as a basis for this research, instead of a vendor specific system, facilitates generalizability.

### **4.5.2 Internal validity**

The main concern regarding internal validity when conducting a case study is the problem of making inferences whenever an event cannot be directly observed. This could be a concern if the researchers are biased (Yin, 2009, p. 43). To reduce making incorrect inferences based on either personal experience, data collected from interviews or internal reports, plausible explanations have been carefully considered and consulted with key informants of NOVN before making an inference.

## 4.5.3 Construct validity

The authors of this master thesis have chosen to follow the recommendations made by Yin (2009, p. 41), and has thus taken several measures to establish construct validity. Multiple sources of evidence have been used as a basis for this research. This is reflected in the amount of reliable scientific journals that have been used to identify operational measures for the concepts being studied. The questions asked in the interviews are grounded in theory. The number of informants that have been selected for interviews also allows the case study findings to be more convincing and accurate as they are based on multiple sources of information. To achieve well-structured interviews, an informant of NOVN carefully reviewed the interview guide before the interviews were conducted. Terms were rephrased to suit the office language of the key informants to make it easier for them to understand the questions. After the interviews were carried out, the informants were again contacted in situations where there was a need to clarify and expand on received information. Corrections were conducted when necessary. The case analysis were also interpreted and discussed in collaboration with key informants. Letting informants within the given unit of NOVN review the draft of the case study report has further increased construct validity (Yin, 2009, p. 41).

# 4.5.4 Reliability

Reliability is important if the research is to be replicated in the future. A case study protocol and database have therefore been developed in order to increase the reliability of the research conducted in this thesis. The former includes the main steps taken to conduct this research, while the latter contains the documentation of the data collected (Yin, 2009, p. 45).

# 4.6 Summary

To answer the two research questions the research of this thesis was designed as a mixed method research using descriptive and exploratory research. By performing an assessment of relevant available literature on the phenomenon in question and by conducting interviews with several key informants, this research was carried out using both the quantitative and qualitative research methods literature review and case study. A summary of key aspect of the research methodology used for answering the two research questions can be found in Table 4.1.

Research question	RQ1: What is the impact of an ATO based PPC system on an ETO company?	RQ2: How can an MRP system be used to support the PPC needs of an ETO company?
Purpose	Identify strength and weakness of the ATO based PPC system and its impact on an ETO company	Increase understanding for how aspects of a used MRP system fit the PPC needs of NOVN
Research design	Descriptive and exploratory	Exploratory
Research methods	Literature review and case study	Literature review and case study
Data collection	Theoretical framework collected from literature based on keywords and backwards literature search, insights from case study such as quantitative data and interviews	Empirical insights such as qualitative data in form of personal- and group interviews and informal conversations, and proposed solutions collected from literature based on keywords and backwards literature search
Data analysis	ANOVA test, Scheffe's test and content analysis	Content analysis

Table 4.1: An overview of the research methodology used in the thesis

# 5 CASE STUDY: NATIONAL OILWELL VARCO NORWAY

This chapter introduces the case study company that has been used to be able to answer the research objective of the thesis.

Section 5.1 gives a brief introduction to NOV and NOVN and how this company can be considered as a representative for ETO companies. Section 5.2 gives a profound review of the unit where the research has been conducted. This section also describes what operations strategy was used and how planning and control was performed in the given unit during series production.

# 5.1 National Oilwell Varco Inc.

National Oilwell Varco Inc. (NOV) is a worldwide leader in the design, manufacture and sale of equipment and components used in oil and gas drilling and production, the provision of oilfield inspection and other services, and in supply chain integration services. This makes NOV a single source provider to the upstream oil and gas industry (NOV, 2013, 2014a).

The company operates in three financial segments: Rig Systems, Wellbore Technologies, and Completion and Production Solutions (NOV, 2014c). Their vision is to be the world's leading and most innovative provider of both drilling and handling equipment as well as related services to the energy industry (NOVN, 2011).

NOV's headquarter is located in Houston, Texas. They have over 63,000 employees working in 60 countries, and had a revenue of US\$22.77 billion in 2013 (NOV, 2014b; NOVN, 2013). Their history dates back to 1841, with the establishment of the company Brissonneau Brothers. Through several mergers and acquisitions NOV has become the worldwide leading company it is today ("NOV Milestones [video clip]," 2014).

## 5.1.1 National Oilwell Varco Norway

In 2003 NOV purchased the Norwegian company Hydralift, which became the start of National Oilwell Varco Norway (NOVN) (Reinertsen, 2013). NOVN is therefore a wholly owned subsidiary of NOV. NOVN is a part of Rig Systems, and their head office is in Kristiansand. The company is also located in other Norwegian cities, such as Stavanger, Molde, Oslo, Tønsberg, Arendal and Trondheim (NOVN, 2012).

NOVN has approximately 4,700 employees, and is Southern Norway's largest private employer (NOVN, 2012, 2013). They were ranked as Norway's 3<sup>rd</sup> largest exporter in 2012 and had a revenue of 31 billion NOK in 2013 (Fjose et al., 2013; Reinertsen, 2014). 75% of all new projects within the industry are carried out by NOVN, and at the end of 2013 NOVN's backlog was approximately 55 billion NOK, which amount to 56% of NOV's entire backlog.

NOVN's expertise is in quoting, engineering, project management, manufacturing, commissioning and field service (NOVN, 2008). A large part of their production is based on intermediate products that are produced in other NOV locations, mainly in USA, but also from countries such as South Korea, China and Poland (Fjose et al., 2013).

## 5.1.1.1 Company objective and performance measures

NOVN's overall company objective is to be the world's leading and most innovative provider of drilling and handling equipment and related services to the energy industry. NOVN operates on the basis of four guiding principles: quality, safety, innovation and growth. While growth refers to the ability to expand in the market NOVN operates in, the safety of a product is considered to be depending on the quality of the product. The two former principles are therefore both related to quality, whilst the level of innovation depends on the flexibility of the company. This therefore suggests that the most important performance measure of NOVN is quality and flexibility.

## 5.1.1.2 Characteristics of NOVN

As NOVN's decoupling point is located at the design stage where the order is linked to the customer, the company is considered to have an ETO operations strategy. For the generalizability of this research it is of paramount importance that the case study is a representative case for ETO companies. NOVN will therefore be described based on the characteristics identified in Section 2.1.3.1.3.

## 5.1.1.2.1 Market related characteristics

As NOVN operates within the oil and gas industry, its customers include drilling contractors, shipyards and other rig fabricators, as well as well servicing companies, national oil companies, major and independent oil and gas companies, and pipe-running service providers.

Although the size of these customers varies, each customer is characterised as large. For example, in 2009 one customer accounted for about 16.6% of NOV's total revenue. Customers of this magnitude constitute a major risk as loosing such customers will lead to a large reduction in revenue. This can be the result if the oil and gas prices fall below a range that is acceptable to the customer, thus resulting in a significant reduction in demand for NOVN's products (NOV Inc., 2010, p. 22). The reason for why the oil and gas prices could have such a large impact on revenues is because drilling activities are closely linked to the fluctuations of oil and gas prices that are determined in the market place. This means that the expectations for future oil and gas prices causes several shifts in the strategies and expenditure levels of oil and gas companies as well as drilling contractors, such as NOVN (NOV Inc., 2006, p. 15). Because of this, the market NOVN operates in can be characterised as highly volatile.

One of the main characteristics of NOVN is its ability to customise products according to customer requirements. As mentioned earlier, this poses a constraint on the company's delivery lead-time. Because of this, NOVN is dependent on early customer involvement already at the design stage. Although some of these customer requirements can be satisfied with previous designs, most orders need to be modified or designed from scratch. This wide range of differences in market requirements is reflected in the extensive product range offered by NOVN. The company's degree of product innovation is therefore mainly driven by new customer requirements.

### 5.1.1.2.2 Product related characteristics

NOVN's product portfolio consists of a wide range of products, all with distinct characteristics. However, as most of NOVN's products demand early involvement of the customer in the production process, they can be characterised as innovations or as products experiencing growth in sales and profit. The overall product portfolio of NOVN can therefore be located in the introduction, or in the growth stage of the PLC (cf. Section 2.1.2.2).

Most products produced by NOVN require a large diversity of components as well as an extensive amount of assembly processes to finalise the product. The general BOM is therefore A-shaped, which means that higher levels consumes the lower levels of raw material or components (Avraham & Karni, 2013). Thus, the products NOVN produce have a deep multileveled product structure.

### 5.1.1.2.3 Production related characteristics

Within NOVN, the enquiry-, design- and engineering stages are considered to add the most value to the company's products. However, because of the frequent engineering changes that occur due to the close involvement of NOVN's customers, these stages are also considered to be the company's main bottlenecks. NOVN's production characteristics therefore underline the unpredictability of these stages.

The above identified market- product- and process characteristics of NOVN are consistent with the characteristics of an ETO company elaborated on in Section 2.1.3.1.3. NOVN is therefore considered to be a representative case of ETO companies.

# 5.2 Roughneck unit

The roughneck unit is project based and has an ETO operations strategy in line with the overall operations strategy of NOVN. The roughneck unit designs several products. One of them is the pipe handling machine roughneck, often referred to as hydratong within NOVN. This machine is used to connect and disconnect pipes on the drill floor and eliminates the manual handling involved with suspended individual tools. It comes in nine different models all with distinctive features. These models consist mainly of components that are commonly used on most of the models, but they can also be adapted to customer requirements. In line with the ETO operations strategy the CODP of the roughneck unit is located at the design stage of the supply chain.

The decision to undergo series production of roughnecks was a result of market shifts. In 2004, the time prior to implementation of the ATO based PPC system, the roughneck unit experienced a significant increase in customer orders. They especially experienced an increased demand for their most sold roughneck model, the MPT-200. As the roughneck unit at the time was delivering roughnecks at full capacity, accepting these orders would result in long delivery lead-times, which the customers would not accept. The roughneck unit acknowledged that the link between the market and production of roughnecks had drifted out of alignment creating a misfit, and that their current ETO strategy was not efficient enough to cater the new characteristics of the market. To cope with the increased demand the roughneck unit started to search for new ways to improve their production, planning and control processes in terms of speed and product quality. This resulted in a new operations strategy for production of the MPT-200 machine: ATO, also referred to as series production within

NOVN. To support the PPC needs of the roughneck units new operations strategy, a customised ATO based PPC system was developed. This system was not a software, but mainly consisted of working manually in Microsoft Excel, and can thus not be found in any literature. The system was fully implemented within the unit early in 2006 and was fully operable until late 2009. At this point it was decided to shut down the ATO based PPC system and return to their original ETO operations strategy. This decision to shut down series production of roughneck in 2009 was taken at the corporate level of NOVN in the wake of a failed attempt of series production of another product within another unit in NOVN. However, no formal evaluation of the ATO based PPC system and its performance has been conducted.

## 5.2.1 Assemble-to-order operations strategy

In line with the ATO operations strategy, the product structure of the roughneck needed to be decomposed into modules to reduce lead-time and increase operations efficiency. The roughneck's many features got broken down into modules that were considered to satisfy most of the current customer requirements. Instead of developing an equipment package for every project, only one equipment package was made for each module and used for several projects. Furthermore, documentation of the components was now issued per module, instead of per projects, as was the case prior to the ATO based PPC system. A module number for every module was created and had its own BOM.

As production lead-time poses a major constraint on delivery lead-time, establishing additional production capacity was crucial to facilitate an ATO operations strategy. Contracting with a Norwegian workshop, as a supplement to the original workshops located in Poland, became necessary in order to increase production capacity and hence be able to respond more quickly to the market. The excess workshop was assigned the most urgent jobs.

To avoid inventory cost and the risk of being left with obsolete items in inventory, purchasing orders were only issued when a contract with a customer was signed. Thus no components were purchased on speculation. In line with the ATO operations strategy identified in Section 2.1.3.1, the ordering of the modules can be seen as forecast driven activities as these are not linked to any project, while the linking of modules to project and final assembly can be seen as customer-order-drive activities. The ATO operations strategy therefore resulted in a forward displacement of the decoupling point, now located at the final assembly stage.

## 5.2.2 The ATO based PPC system

The ATO based PPC system was designed to provide an overview of the production of the modules as well as the pending projects waiting to be linked to its respective modules. When the roughneck unit received orders of MPT-200, they were placed into the ATO based PPC system. Based on the customer requirements, the appropriate modules for the project was identified and calculation of the material requirements was done by exploding these modules BOM's. If the customer requirements consisted of specifications beyond the modules, these requirements were handled separately and ordered directly on the specific project number. Instead of ordering components on specific project numbers on an on-going basis as the unit used to do prior to the ATO based PPC system, the modularity of the MPT-200, and the fact that documentation was attached to every component, allowed for ordering batches of components. Furthermore, the modularity of the MPT-200 allowed purchase orders to be issued more quickly than before, as there already existed an equipment package for the module. The components were therefore interchangeable before they were tied to a specific order at the CODP. The production progress of each module was reported in the ATO based PPC system so that it was easy to keep an overview. When the modules were completed, they were linked to the project that was most urgent to complete. The specific module used to build a specific roughneck was added to the general assembly drawing of the given project, which facilitated full traceability of each component, as well as its documentation, in every module, and hence the project. The cost of the projects was transferred to the project number and neutralised at the module number when the project was completed and delivered to the customer.

At the time of series production, and thus using the ATO based PPC system, the roughneck unit consisted of five employees. Two employees were dedicated workers on series production: a production manager and a technical assistant. As a roughneck consists of necessary components supplied by the electrical and hydraulic department, a contact person from each of these departments was appointed to be dedicated to series production of roughnecks and in charge of defining necessary hydraulic and electrical components. Additionally, a person within the purchasing department was also appointed to handle the purchase orders that were placed in regard to series production.

The production manager had the overall responsibility of managing the system and served as a single point of contact in relation to the workshops, the electrical and hydraulic department, as well as the purchasing- and cost and controlling department. The production manager thus was responsible for registering orders, purchase orders, notifying the different departments of new orders and follow up production as well as being responsible for keeping the ATO based PPC system updated at all times. The technical assistant assisted in this work. The remaining three employees were equipment responsible (ER) and were in charge of the total project delivery. Their main focus was on the technical aspects of the project and additional specifications that customers required and that was not produced by performing series production. The ER's had full insight into the system<sup>2</sup>.

<sup>&</sup>lt;sup>2</sup> Source of information is collected from the conducted interviews.

# 6 RQ1: EVALUATION OF THE ATO BASED PPC SYSTEM

In this chapter the findings related to the quantitative and qualitative evaluation of the ATO based PPC system are presented and discussed, thus answering RQ1.

Section 6.1 briefly presents what forms the basis of the quantitative and qualitative analysis. Section 6.2 analyse and evaluates the quantitative performance measures of the ATO based PPC system. Section 6.3 analyse and evaluates a number of subjective measures and provides a brief summary of the overall findings of RQ1.

# 6.1 Quantitative and qualitative evaluation

The performance of a PPC system is affected by how it is designed as well as numerous external conditions and circumstances that is beyond the company's control. To evaluate the ATO based PPC system implemented in the roughneck unit, a quantitative and qualitative analysis has been conducted. The analysis is based on the operational measures of the performance measures and the subjective measures identified in Section 2.1.2.1 and elaborated on in Section 2.3. In some cases performance data could be extracted directly from NOVN's information systems, however most of the analysis required collecting, structuring and cleansing large amounts of data. The statistics on the operational performance measures of the roughneck unit form the basis of the quantitative analysis; the quasi experiment. The findings from two interviews conducted with three key informants that were employed in the roughneck unit during the use of the ATO based PPC system provides the basis of the qualitative analysis. Parts of the findings of this chapter are therefore based on the authors' interpretations of evidence from these two interviews.

# 6.2 Quantitative measures

As explained in Chapter 3, six hypotheses have been formulated based on the tangible measures; quality, speed, dependability, flexibility, cost and value added:

H<sub>1</sub>: The ATO based PPC system led to a reduction in unforeseen extra working hours.
H<sub>2</sub>: The ATO based PPC system led to a reduction in delivery lead-time.
H<sub>3</sub>: The ATO based PPC system led to reductions in frequency of imperfect deliveries.
H<sub>4</sub>: The ATO based PPC system led to a decrease in working hours per roughneck.

*H*<sub>5</sub>: The ATO based PPC system led to a reduction in costs per roughneck.*H*<sub>6</sub>: The ATO based PPC system led to an increase in profit margin.

The relationships hypothesised are depicted in Figure 6.1.



Figure 6.1: Research model.

These hypotheses will be tested by conducting a quasi-experiment of the ATO based PPC system's impact on performance measures. The quasi experiment is designed as followed:

(01 X 02) 03
01 = Pre-period
02 = Series production
03 = Post-period
X = ATO based PPC system

The analysis is based on data on MPT-200 deliveries during the period of series production (02), lasting from 2006 until 2009. Equivalent data on the period prior to series production, 2004–2006, makes up the pre-period (01). Similarly, data on the period after series production, 2009–2012, constitutes the post-period (03). The post-period serves as a control period to establish that the observed effects are due to the independent variable; the ATO based PPC system (X). External effects and maturity effects that are considered to have an explanatory power of the observed results have been identified and taken into consideration when

interpreting the results. The results of the ANOVA and Scheffe's test are presented in the forthcoming sections, followed by an interpretation of the result.

# 6.2.1 Quality

As explained in Section 3.2, failure costs have been identified as an appropriate measure for evaluating what impact the ATO based PPC system has had on product quality. Unforeseen extra working hours, which is a common measure of internal failure cost within NOVN, will reflect the systems impact on product quality. Such hours often occur because a deviation from the standard design quality has occurred. This deviation must then be corrected before the product is transferred to the customer. An example of such a deviation is defective products or materials that cannot be repaired, used or sold. The amount of unforeseen extra hours that have elapsed on the roughneck projects has therefore been measured in the three periods studied.



The overview of the average amount of unforeseen working hours of the three periods is depicted in the above diagram. Compared to the pre-period, the roughneck unit experienced a decrease from approximately 70 hours to 63 hours during series production. Unforeseen

amount of working hours was further reduced to 20 in the post-period.

The null hypothesis in the ANOVA test:

 $H_0$ : There exist no significant differences in variances in reduction of unforeseen extra working hours between the three periods measured.

### ANOVA

Unforeseen working hours

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	72370.972	2	36185.486	1.844	.161
Within Groups	3277431.434	167	19625.338		
Total	3349802.406	169			

Critical value of F = 3.050

The analysis of variances of the three different periods is shown in the table above. The F-value of the test is 1.844 and the critical value of F is 3.050. As the F-value for this test does not exceed the critical value (1.844 < 3.050) the null hypothesis cannot be discarded. This result is reflected in a significance level well above 0.05 (0.161 > 0.005). It is therefore not possible to say with 95% certainty that there exist significant variances of unforeseen working hours for the different periods. The ANOVA test does therefore not provide any support for the hypothesis stated in H<sub>1</sub>, but supports the H<sub>0</sub>; the implementation of the ATO based PPC system did not lead to a reduction in unforeseen extra working hours.

# 6.2.2 Speed

In Section 2.3 delivery lead-time was identified as an appropriate measure to investigate if the ATO based PPC system has had an impact on the ability of the roughneck unit to quickly respond to demand signals. The delivery lead-time on roughneck projects has therefore been calculated for the time elapsed from the contract sign date to the contract delivery date.



As can be seen from the overview of the average delivery lead-time of the three periods depicted in the diagram above, the delivery lead-time has had a steady decline over the three periods.

## The null hypothesis in the ANOVA test:

 $H_0$ : There exist no significant differences in variances in delivery lead-time between the three periods measured.

### ANOVA

Delivery lead-time

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	1157921.176	2	578960.588	22.195	.000
Within Groups	1852075.316	71	26085.568		
Total	3009996.491	73			

Critical value of F = 3.125

As can be seen from the table, the F-value of the test exceeds the critical value of F (22.195 > 3.125) and the significance level is well below 0.05 (0.00 < 0.05). The variance analysis

therefore gives a strong indication that there are significant variances between the three periods that have been measured.

A Scheffe's test must be conducted to establish what periods contribute to these variances.

The hypothesis to be tested in the following Scheffe's test: *H*<sub>2</sub>: *The ATO based PPC system led to reductions in delivery lead-time*.

#### **Multiple Comparisons**

Delivery lead-time

Scheffe

(I) Period	(J) Period	Mean Difference (I-J)	Std. Error	Sig.
Pre-period	Series production	219.171	88.850	.054
	Post-period	432.123 <sup>*</sup>	83.862	.000
Series production	Pre-period	-219.171	88.850	.054
	Post-period	212.951 <sup>*</sup>	43.410	.000
Post-period	Pre-period	-432.123 <sup>*</sup>	83.862	.000
	Series production	-212.951 <sup>*</sup>	43.410	.000

\*. The mean difference is significant at the 0.025 level.

As can be seen from the table above, the pre-period had significantly longer lead-times than the period of series production. The Scheffe's test provides strong evidence, with a significance level of 0.05, which support the hypothesis stated above; the implementation of ATO based PPC system led to reductions in lead-time.

However, Scheffe's test also shows that the roughneck unit experienced similar reduction in delivery lead-times when comparing the pre- and post-period. It therefore indicates a

maturation effect, which can have an impact on the observed result, thus reducing the likelihood that the ATO based PPC system has had a positive impact on delivery lead-times of roughnecks.

## 6.2.3 Dependability

The frequency of late or imperfect deliveries is the identified measure that can reflect if the ATO based PPC system has had an impact on delivery reliability of the roughneck unit, as explained in Section 2.3. The frequency of imperfect deliveries has therefore been measured across the three periods of time.



An overview of the average delayed projects is depicted in the diagram above. While all projects in the pre-period were delayed, approximately half of the project was delayed during the series production and post-period.

### The null hypothesis in the ANOVA test:

 $H_0$ : There exist no significant differences in variances in frequency of imperfect deliveries between the three periods measured.

## ANOVA

Projects delayed

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	2.390	2	1.195	5.001	.008
Within Groups	39.899	167	.239		
Total	42.288	169			

Critical value of F = 3.050

The result of the ANOVA test shows a F-value larger than the critical value, and a significance level well below 0.05. This provides strong evidence for that there exist significant differences in variances between the average of delayed projects across the three periods. The null hypothesis can therefore be discarded.

To establish what periods contribute to these variances, it is necessary to conduct a Scheffe's-test.

The hypothesis to be tested in the following Scheffe's test:

*H*<sub>3</sub>: *The ATO based PPC system lead to reductions in frequency of imperfect deliveries.* 

#### **Multiple Comparisons**

Projects delayed

Scheffe

(I) Period	(J) Period	Mean Difference (I-J)	Std. Error	Sig.
Pre-period	Series production	.477*	.162	.014
	Post-period	.529 <sup>*</sup>	.169	.009
Series production	Pre-period	477 <sup>*</sup>	.162	.014
	Post-period	.052	.083	.820
Post-period	Pre-period	529 <sup>*</sup>	.169	.009
	Series production	052	.083	.820

\*. The mean difference is significant at the 0.025 level.

The Scheffe-test unveils that there are significantly fewer delayed projects during series production, compared to the pre-period. This analysis strongly suggests that the ATO based PPC system had a significant positive impact on the amounts of delayed projects, supporting the hypothesis stated above.

The test also shows that a small, but significant improvement in projects delays was achieved when comparing the pre- and post-period. This indicates maturation effect, which again reduces the likelihood of the results observed above.

## 6.2.4 Flexibility

As explained in Section 5.2.2, the components were interchangeable before they were tied to a specific order at the CODP. The ATO based PPC system should therefore enable the roughneck unit to respond to changes in for instance a delay in delivery lead-time for a component as identical components scheduled for other projects could be allocated to the more urgent projects. In Section 2.3, working hours per roughneck have been identified as an appropriate measure which can give an indication of the flexibility of operations within the roughneck unit, as a decrease in hours could indicate excess capacity and hence flexibility. To measure if the ATO based PPC system led to a decrease in working hours per roughneck, the amount of hours elapsed per roughneck have been measured across the three periods of time.



As can be seen from the overview of the average amount of hours elapsed per roughneck provided above, the roughneck unit experienced an increase of one hundred hours during series production, compared to the pre-period. During the post-period the amount of hours decreased by two hundred, compared to when series production was used.

## The null hypothesis in the following ANOVA test:

 $H_0$ : There exists no significant variance in frequency of imperfect deliveries between the three periods measured.

### ANOVA

Working hours

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	1417888.662	2	708944.331	1.847	.161
Within Groups	6.409E7	167	383756.863		
Total	6.551E7	169			

Critical value of F = 3.050

The variance of analysis has an F-value below the critical value of F and therefore shows a weak result of significant differences between the periods in terms of amount of hours elapsed per roughneck, with a significance level well above 0.05. The null hypothesis can therefore not be discarded as the test does not provide an indication of that there exist significant differences between the three measured periods. Hence, the test does not provide any support to  $H_4$  stating that the ATO based PPC system has led to increased flexibility.

Findings of the interviews however indicate that the roughneck unit had considerable higher flexibility during series production, compared to the pre- and post-period. The three informants employed in the roughneck unit at the time of series production all state that they experienced less workload during series production, compared to the other two periods. As one key informant explained:

Even though most work tasks had not changed, we were able to accumulate the tasks and execute them for several projects at once.

This suggests that part of the reduction in workload was facilitated by the ATO based PPC system, which made it easier for employees within the roughneck unit to execute their tasks more efficiently. Furthermore, as one key informant explained:

During series production it was possible to have a single point of contact responsible for communication with workshops. This single point of contact could take over many of the task that the ER was previously responsible for.

This suggests that the use of a single point of contact alleviated the amount of work the ER conducted in relation to purchasing of activities and communication with the workshops. During series production the ER's could therefore have a more technical focus on the completion of the projects.

Another evidence of the flexibility during series productions compared to the other periods is stated by one of the key informants:

During series production we managed with less ER's than we do today, as every ER handled more projects simultaneously.

While the roughneck unit managed with only three ER's during series production, it consists of six ER's in the post-period. This suggests that the roughneck unit increased their excess capacity making them more flexible in response to changes in demand during series production. However, the key informants state that part of this reduction in flexibility could be attributed to the increase in specific customer requirements per order and the increase in product variety. This made it more difficult for the ER's to keep an overview of what was ordered on each project.

The key informants state that the greatest flexibility was achieved by ordering components on module numbers, instead of project numbers, and the fact that the appropriate documentation was attached to every component. This allowed the roughneck unit to swap the modules between the projects, making the unit highly flexible when customers changed their requirements during the project, when the unit accepted an urgent order, or in case of a breakdown. The modules were then swapped to the project which required the specific module or that was most urgent to complete.

## 6.2.5 Cost

To measure if the ATO based PPC system has had a positive impact on cost in roughneck production, cost per roughneck have been measured.


The above diagram provides an overview of the average cost per roughneck during the three different periods. As can be seen, a two times increase in cost per roughneck occurred during series production, compared to the pre-period, and remained stable throughout the post-period.

#### The null hypothesis in the following ANOVA test:

 $H_0$ : There exist no significant differences in variances of cost per roughneck between the three periods measured.

#### ANOVA

Cost

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	5.322E11	2	2.661E11	4.274	.015
Within Groups	1.040E13	167	6.226E10		
Total	1.093E13	169			

Critical value of F = 3.050

The analysis of the periods measured show a strong result of that there exist significant differences in variances between the different periods in terms of cost per roughneck.

To establish what periods contribute to these variances, it is necessary to conduct a Scheffe's-test.

The hypothesis to be tested in the following Scheffe's test: *H*<sub>5</sub>: *The ATO based PPC system led to a reduction in costs per roughneck.* 

#### **Multiple Comparisons**

Cost

Scheffe

	-			
(I) Period	(J) Period	Mean Difference (I-J)	Std. Error	Sig.
Pre-period	Series production	-234530.301 <sup>*</sup>	82446.488	.019
	Post-period	-186612.475	86296.217	.100
Series production	Pre-period	234530.301 <sup>*</sup>	82446.488	.019
	Post-period	47917.826	42332.468	.528
Post-period	Pre-period	186612.475	86296.217	.100
	Series production	-47917.826	42332.468	.528

\*. The mean difference is significant at the 0.025 level.

The result of the Scheffe's test reflects a considerable increase in cost between the pre-period and the period of series production. A significance level of 0.01 is a strong indication of that the ATO based PPC system has led to increased costs per roughneck.

Furthermore, the Scheffe's test shows that there are no significant changes between the period of series production and the post-period. This indicates that no maturity effects exist, which reduce the likelihood of the result derived above.

### 6.2.6 Value added

Value added is a measure of profitability that can be reflected through the profit margin. To analyse the roughneck unit's ability to achieve a greater sense of value of their MPT-200, as perceived by their customers, the average profit margin has been measured for the three periods of time.



An overview of the profit margin across the different periods is provided in the diagram above. While the profit margin was roughly fifteen percent during the pre-period, it was over three times as large during series production. In the post-period, the profit margin decreased by seven percent.

The null hypothesis in the following ANOVA test:

 $H_0$ : There exist no significant differences in variances of profit margin cost between the three periods measured.

#### ANOVA

Profit margin (%)

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	10337.183	2	5168.592	21.469	.000
Within Groups	40203.880	167	240.742		
Total	50541.064	169			

Critical value of F = 3.050

As can be seen from the above table, a F-value well above the critical value of F suggests that the null hypothesis should be discarded, as there exist significant differences between the different analysed periods.

To establish what periods contribute to these variances, it is necessary to conduct a Scheffe's-test.

The hypothesis to be tested in the following Scheffe's test: *H*<sub>6</sub>: *The ATO based PPC system led to an increase in profit margin.* 

#### **Multiple Comparisons**

Profit margin (%)

Scheffe

(I) Period	(J) Period	Mean Difference (I-J)	Std. Error	Sig.	
Pre-period	Series production	-32.61583 <sup>*</sup>	5.12668	.000	
	Post-period	-25.79871 <sup>*</sup>	5.36606	.000	
Series production	Pre-period	32.61583 <sup>*</sup>	5.12668	.000	
	Post-period	6.81713	2.63231	.037	
Post-period	Pre-period	25.79871 <sup>*</sup>	5.36606	.000	
	Series production	-6.81713	2.63231	.037	

\*. The mean difference is significant at the 0.025 level.

The Scheffe-test strongly indicates, with a significance level well below 0.05, that there are significant differences in profit margins obtained between these periods. The result show that there is a clear indication that the ATO based PPC system has led to an increase in profit margin of the roughneck unit.

There is a sign of maturity effect, as the test show that the roughneck unit experienced significantly higher profit margin in the post-period, compared to the pre-period. This indicates that there exists maturation effect that decreases the likelihood of the result that has been derived above.

The result of the quasi experiment and its supported hypotheses are summarised in the figure below:



Figure 6.2: Research model with significant impacts

Findings of the interviews do not support all the results from the quantitative analysis. In terms of cost per roughneck, two of the key informants claim that the roughneck unit experienced cost reduction during series production due to ordering in larger quantities, which is not in line with the result of the conducted analysis. This indicates that the key informants have an incorrect perception of the performance of the ATO based PPC system in terms of cost. The key informants claim that there was an increase in fast track projects during series production. It is therefore appropriate to make the inference that part of the increase in contract prices during the period of series production can be attributed to the increased amount of fast track projects. As the increased flexibility of the roughneck unit's production capacity allowed them to take on more urgent projects, short delivery lead-time led to additional value added. This resulted in higher contract prices, which the customer was willing to pay. The key informants also confirm that the profitability went down after series production was shut down, and that this reduction was related to an increase in special requirements per order. This had significant implications on operations, and furthermore it limited the possibility to order components in large quantities and thus reduce cost.

In sum, the ATO based PPC system is considered to have had a significant positive impact on the performance of the roughneck unit. However, the results of the quasi experiment show that the improvements experienced in terms of quality and flexibility are not significant enough to attribute them to the success of the ATO based PPC system. Even though, the findings of the interviews suggest that the roughneck unit achieved greater flexibility during the period of series production. The result indicates that the ATO based PPC system has had significant positive impact on dependability, cost and value added. In terms on dependability, delivery reliability was reduced by nearly fifty percent during series production. In terms of cost, the result indicates that despite the fact that cost per roughneck increased significantly during series production, a larger increase in value added resulted in higher contract prices and thus higher profit margins. Only a small reduction in lead-times was achieved. It therefore remains unclear whether this reduction should be attributed to the ATO based PPC system or not.

## 6.3 Qualitative measures

During the collection of data a number of qualitative aspects, elaborated on in Section 2.3, was measured to provide a better understanding of the performance of the ATO based PPC system. The goals of the new system, as well as education of personnel and gaining acceptance and support, will therefore be evaluated based on findings from the interviews. As no raw material or components were ordered on speculation, forecasting and accuracy is not applicable to the evaluation during the period of series production.

### 6.3.1 Education of personnel

Both ahead and during start-up with the roughneck unit's employees, several meetings were held and formal procedures and guidelines for use of the ATO based PPC system were developed. However, the general interpretation of the employees employed in the roughneck unit during the use of the ATO based PPC system is that no formal training program was set up for the users of the system when the system was implemented. As one of the key informants explained:

A "learning by doing" approach was applied for education of the system (...) it was not organised systematically.

The three interviewed key informants that worked in the roughneck unit at the time of the ATO based PPC system claim that the time dedicated to education and the level of training

they received was adequate. Even so, two of the key informants pointed out that it could have been organised more systematically.

## 6.3.2 Goals

There is some disagreement as to what the goals of the ATO based PPC system was and how well the goals were communicated from the head of the roughneck unit to the employees. While one of the key informants claim that the goals of the system were well communicated to other colleagues, the remaining two key informants could not render the specific goals of the system. This suggests that there was a lack of communication between the head of the unit and its employees.

## 6.3.3 Support

The general interpretation is that most disciplines and support activities that were affected by series production, and whose support the roughneck unit was dependent on, found it difficult to adapt to the new ATO based PPC system. As one of the key informants put it:

As this was a new system, it was hard for the other disciplines and support activities to adapt to it.

The hydraulic- and electrical department were at first reluctant to adapt to the new system as the new system required a dedicated employee within the departments to handle the hydraulics and electrical components of roughnecks produced in series. Other departments facilitating support activities, such as inventory-, cost and controlling-, and the purchasing department, also found it difficult to adapt to the new system. This was because the system forced these departments to change their routines to accommodate the roughneck unit. As one of the key informants explained:

The inventory department found the new system difficult to adapt to. This was due to the fact that instead of grouping all components to one specific project number, as they used to do earlier, the components were grouped into boxes per module. Inventory therefore had to manage considerably more boxes, before the different modules eventually got linked to a specific project number. The cost and controlling department on the other hand found the new system to be problematic as they were not able to transfer the cost from the modules to the project before it was delivered to the customer. And as modules for the projects were ordered early in the project phases it lead to increased inventory costs. The fact that the department was not able to immediately expense these inventory costs to a project, contributed to their frustration with the ATO based PPC system. The purchasing department also found the new system somehow difficult. They often approached the ER's at the roughneck unit to check if the necessary components had been ordered, as the purchase orders were not linked to a project number. These other departments simply could not understand why the roughneck unit needed to do things differently than the rest of the company.

#### 6.3.4 Gaining acceptance

When it comes to the new ATO based PPC system gaining acceptance, the employees within the roughneck unit seemed to embrace the changes necessitated by the system relatively fast. While two key informants claim there were no employees that showed any signs of reluctance, the third claimed that those who appeared to be reluctant at first, quickly saw the benefits of the system and ended up embracing the system as well.

The evaluation of the roughneck unit during series production have resulted in a number of key findings with regards to the way production of roughnecks was planned and controlled during the time of the ATO based PPC system. Overall, the quantitative analysis complimented by the findings of the interviews, largely supports that the ATO based PPC system have in fact had an effect on the performance of the roughneck unit. The quantitative analysis concludes that the most significant changes experienced were in terms of increased dependability and profitability of the roughneck unit. However, the quantitative analysis does not conclude that the ATO based PPC system supported the overall objective of the ATO operations strategy, which should become evident by an increase in speed and product quality. The qualitative aspects analysed uncovered some weaknesses of the ATO based PPC system and how it was implemented. The main findings was that the goals of the ATO based PPC system were not communicated well to the employees of the roughneck unit and that there was a lack of support from other departments within NOVN. This is in line with the most frequent problems encountered during PPC system implementation, as explained in Section 2.3. There is reason to believe that these problems have had a negative impact on the performance of the ATO based PPC system. In sum, it can be concluded that the roughneck

unit's ATO operations strategy, in comparison to the current ETO operations strategy of NOVN, appeared to be more effective in meeting customer requirements in terms of dependability and value added. However, it did not sufficiently support the overall objective of the operations strategy. Therefore, the ATO operations strategy did not successfully serve as the interface between the overall objective and operations.

## 7 RQ2: MRP IN AN ETO ENVIRONMENT

As mentioned in Section 2.4, an MRP system is largely constrained by a company's product-, market- and process characteristics. These characteristics will therefore influence how well the MRP system is able to accommodate the PPC needs of NOVN, and hence the implementation of an MRP system into an ETO company, such as NOVN. This chapter gives an extensive review on the applicability of a MRP system within NOVN, thus answering RQ2.

The stages that add most value to NOVN's customers, and that are considered to be NOVN's competitive advantage, are the enquiry-, design- and engineering stages. As mentioned in Section 5.1.1.2.3, these stages are also considered to be NOVN's main bottlenecks, and they are therefore subject to PPC. Information regarding cost of inputs, processing cost and contract price are considered to be of paramount importance for NOVN to be able to execute PPC of its operations. In respect to this, the PPC needs of NOVN throughout these stages have been identified and explained. An analysis of how an MRP system will accommodate these and further support NOVN's operations strategy will follow. Lastly, the MRP's effect on the company objectives of NOVN will be discussed.

Ideally, the new PPC solution should have been implemented as part of this research so that the impact of the MRP system could have been measured and verified. Unfortunately such a research strategy was not possible within the available time of the study. Instead, expected impact was substantiated through a combination of logic reasoning and joint reflections with key informants.

#### Incomplete BOM's

The product-, market- and process characteristic of NOVN makes NOVN highly dependent on operating with incomplete BOM's throughout project lifecycles. The importance of incomplete BOM's was further underlined by one informant, which predicted it to become the biggest challenge when implementing the MRP system. There are several factors that make NOVN dependent on operating with incomplete BOM's, and these will now be explained.

The high degree of customisation opportunities offered by NOVN, as well as their early involvement with the customer will lead to frequent engineering changes. This means that the

BOM need to be frequently updated and that it tends to not be complete before in the later stages of the project.

As NOVN 's products often consist of a combination of standard components and components that are customised, specific material with long lead-time is often necessary to purchase in the early stages of a project in order to reduce lead-time and stay competitive in the market. NOVN is therefore dependent on building the BOM while engineering is conducted, and for it to be released gradually with partial information so that engineering, purchase and production orders for long-lead items and components can be issued while others are still being defined. However, BOM's at the early stages of a project tend to be inaccurate as the specifications and scope of an order is still uncertain.

One of the assumptions of the MRP system mentioned in Section 2.4.4 is that the file data, which consist of the input into the MRP system, is accurate and complete to facilitate full traceability of a products items. In the case of NOVN, as explained in the previous paragraphs, this will not be fulfilled until the later stages of the project. Although it is possible to release several BOM's in the MRP system as specifications of the customer order becomes gradually known, the items cannot be reconciled back into one BOM. Incomplete BOM's will therefore not provide full traceability. Another important concern is that when a project progresses, the BOM is continuously updated. Such frequent updates of the BOM will lead to MRP system nervousness, as mentioned in Section 2.4.6.

#### Non-physical processes

Within NOVN there are various projects that are run simultaneously, all at different stages of completion. To monitor the development of a project it is therefore important that all processes can be measured in terms of progress. Processes within enquiry-, design- and engineering stages in NOVN are of non-physical character. While physical processes are easy to measure, the partly creative non-physical processes makes it is difficult to distinguish production phases similar to the physical processes. The fact that products are customised to customer requirements and necessary processes tend to be different for every project, complicates this matter even further. An MRP system cannot cope with non-physical activities, and is therefore not able to accommodate the need for determining the progress of projects.

Non-physical processes have implications for cost allocation as well. Overview of actual cost and resources per project is considered to be important information for NOVN as it provides the basis for future contract prices for similar projects, measure company performance, summarizes contract information and provides the basis for cost reduction programmes. For every project within NOVN a cost budget is made followed by an actual cost budget after the project is completed. However, the MRP system is not able to link for instance production hours and used material to a given project. Thus the MRP system does not facilitate an accurate computation of the actual cost of projects, and this will therefore need to be performed manually.

#### Continuous processes

Due to the product characteristics of NOVN, customer orders are handled as projects. Its production processes are therefore considered to be continuously interdependent on each other. For instance, the design of a product may appear inadequate during the assembly process, and therefore needs to be modified before it can be furthered assembled. However, one of the assumptions of the MRP system is that processes are independent from each other and can be completed on its own. The MRP system will therefore be constrained by the processes of NOVN, as it will not be able to accurately calculate when a component is needed.

#### Non-interchangeable items

While an MRP system regards every item with identical item number as interchangeable between customer orders, this is not straightforward within NOVN. For instance, when there is a breakdown of a component on an urgent project, it is not necessarily straightforward to consume identical components of another project. This is due to the fact that every component has its own set of documentation and is linked to a specific project. The MRP system however will treat these items as being identical, which will lead to the MRP system reporting incorrect inventory status. Furthermore, it may lead the MRP system to not issue necessary purchase orders, as the system will report that the item is already in stock.

#### Uncertainty in supply

There is considered to be high uncertainty in supply of critical components, such as long lead items necessary to finalise NOVN's projects. The data relating to lead-times and production times of components supplied to NOVN are therefore based upon estimates. Because of this, the unpredictability of the market often leads to variances between planned and actual lead-

times. In the MRP system lead-times for each component is considered to be deterministic, and the uncertainty is therefore not accommodated by the MRP system. Wrongful predictions will therefore lead to either excess inventory or longer lead-times for the final product. Furthermore, frequent changes in lead-times within NOVN will lead to frequent updates of the MRP system, which in turn will result in MRP system nervousness.

#### Uncertainty in capacity

When NOVN receives a customer order, specifications and scope is only partially known, and will become more certain as the project progress. As required capacity is determined by the order specifications, the required capacity to fulfil a customer order remains uncertain as well. A corresponding PPC system must therefore facilitate capacity planning under uncertainty. Furthermore, a relative high fraction of the available engineering capacity is applied for quotation preparation at the customer enquiry stage. This may interfere with the amount of engineering capacity required for realising the already placed customer orders. This makes available production capacity for current orders uncertain. However, the MRP system does not take this variation in capacity into consideration, thus leading to either a clogged up production chain, or excess capacity.

#### Lot-sizing technique

As the customer orders of NOVN are one-of-a-kind, inventory are mainly directly related these customer orders. Because of this, NOVN does not keep stock of materials and components as the risk of inventory becoming obsolete is considered to be too high. NOVN therefore applies the lot-for-lot technique, which minimizes the holding costs and reacts to all schedule changes in an MRP system, as mentioned in Section 2.4.6. Because of this, the lot-for-lot technique increases MRP system nervousness, and will thus not be applicable to an MRP system.

From the above discussion it is clear that many of the MRP drawbacks identified in Section 2.4.6 will become evident when implemented in an ETO company such as NOVN. The most evident drawback is the MRP system nervousness, which relates to the main feature of an MRP system, namely the rescheduling function. Although this function is an important technique for quickly responding to the market, its associated nervousness can destroy well-planned MRP schedules.

Now that the PPC needs of NOVN and the MRP system ability to accommodate them have been explained, the system's impact on NOVN performance will be evaluated. Flexibility and quality have been identified as the most important company objectives of NOVN, which are also among the identified benefits of an MRP system. A successful implementation should therefore maintain or improve NOVN's level of flexibility and quality. This flexibility is enabled by the rescheduling effect of an MRP system; as the MRP system easily react and adapt to changes in the market place by rescheduling material requirements for a specific change in a customer order that it provides. However, the high supply and demand uncertainties, as well as the product characteristics of an ETO company such as NOVN, is expected to lead to too frequent updates of the system. This will further lead to MRP system nervousness, which is expected to inhibit the flexibility of NOVN's operations strategy. In terms of product quality, the MRP system is not expected to have direct impact on this company objective, as the system is only concerned with providing the right part, and to meet the schedules for completed products. However, the MRP system is expected to free up resources, which in turn can be redirected to improve product quality. In respect to NOVN, the set of actions required to implement the changes prompted by the MRP system followed by frequent updates of the system, are expected to be too comprehensive and demand too much resources, and thus consume any eventual freed up resources. The MRP system is therefore expected to weaken the overall performance of NOVN in terms of flexibility and quality.

#### Other factors affecting implementation of MRP system

The education level of employees is expected to have an effect on the implementation of an MRP system. Most of the workforce of NOVN consists of employees with higher education, especially within engineering. In this regard, key informants claim that NOVN's highly skilled workforce have a need to unfold themselves creatively by being able to perform product innovation, regardless of whether it is required in order to meet customer requirements or not. Such product innovation leads to frequent engineering changes. The education level of NOVN's employees can therefore inhibit a successful implementation of the MRP system, as frequent engineering changes leads to MRP system nervousness.

The education level of NOVN's workforce is also expected to have an impact on the MRP system being able to gain acceptance of the employees. Information gained in the interviews indicates that employees will be reluctant to adapt to the new system. This is further

complicated by the high education level of employees, which suggests that NOVN's workforce is autonomous. It is therefore expected that employees of NOVN will find it difficult to adapt to a rigid system, like an MRP system, as such a system has lower requirements for autonomy of the workforce.

As previously mentioned in Section 2.4.7, company size may have an impact on the company's expectations of the MRP system. As NOVN is considered to be a large company, it is reasonable to believe that NOVN has higher expectations of the MRP system than smaller companies. In the conducted interviews with key informants of NOVN it became clear that the informants' expectations to the new MRP system are that the new system will contribute to making it easier to keep an overview of multiple projects simultaneously. The interviews also revealed that the system will make it easier for the employees to do their job more efficiently. Considering that several literature studies (cf. Table 2.2) have found MRP systems to be non-applicable to an ETO company, this suggests that the result of the implementation will result in large gaps between the NOVN's expectations and the MRP systems actual performance.

NOVN's employee's previous experience with similar PPC systems is also expected to have an impact on the implementation of the MRP system. The shutdown of series production of roughnecks is well known among the employees of NOVN. It is therefore reason to believe that these incidents can lead to the employees having negative attitudes towards the new MRP system, or any new system for that matter. A successful implementation requires employees' competence and willingness to change, and employees are more prone to support changes if they are ready to make these changes and if they believe in them.

One informant underlines the importance of culture in relation to implementation of an MRP system. He further explains that his experience with NOVN's colleagues in the US tells him that Americans have very different work attitudes compared to Norwegians. Culture and its impact on MRP implementation will therefore be further elaborated in Chapter 8.

From the above discussion it is clear that product-, market-, and process characteristics is too much of a constraining factor to the MRP system; The PPC needs in the continuous cycle of engineering as well as the need for updating BOM's and recalculating schedules within NOVN is not accommodated by an MRP system and thus makes the system prone to MRP system nervousness. MRP system nervousness will further lead to the system becoming inflexible. Furthermore the frequency of updates and the set of actions required to implement these changes are too comprehensive and could therefore lead to fluctuating capacity utilisation resulting in costly instability. It is therefore a clear misalignment between the decision support provided by MRP systems and the decision support required by NOVN at the customer enquiry-, design - and engineering stages, which are the stages within NOVN that are regarded as their competitive advantage. Other factors, such as the education level of NOVN's employees, company size and culture, which will be elaborated on in the following chapter, has also been identified as factors that can have a negative impact on implementation of the MRP system. This clearly indicates that the MRP system will not function optimally in an ETO company, which is consistent with findings from the literature study in Table 2.2.

## 8 ADDITIONAL ELEMENTS TO CONSIDER WHEN IMPLEMENTING AN MRP SYSTEM

During one of the interviews a key informant answered culture, when asked what is the biggest difference between NOV in the US and NOVN. The informant further elaborated this by saying that the two countries have different work attitudes. Because of this response it was decided to further investigate different aspects impact on MRP implementation, with a particular focus on culture.

Section 8.1 describes the need for considering culture differences when implementing an MRP system. Section 8.2 elaborates on why it is important for NOVN to take culture into consideration when implementing the new MRP system. Section 8.3 concludes the chapter and elaborates on other aspects that are important for NOVN to take into consideration when implementing the new MRP system.

## 8.1 Cultures influence on MRP implementation

Several authors argue that cultural preferences intensify MRP implementation problems as companies who implement these types of systems suffer a major change in their organisational information system (Davison, 2002; Olson et al., 2005; Sheu et al., 2004). Through a number of studies conducted on culture and its effect on MRP implementation the conclusion has been that it is necessary to take culture into account when implementing MRP systems in different countries as culture has a significant influence on MRP implementation (Davison, 2002; Hong & Kim, 2002; Olson et al., 2005; Sheu et al., 2004; Soh et al., 2000). Because earlier studies have identified culture as one of the most important factors in determining an MRP systems success or failure, and because research have shown that failure to adapt the MRP system to fit a company's culture can lead to large cost overruns and underperformance (Krumbholz & Maiden, 2001; Leidner & Kayworth, 2006; Sheu et al., 2004), the authors of this master thesis believes it is important for NOVN to focus on culture when implementing the new MRP system. This importance is further amplified as the MRP system that NOVN is going to implement is developed in America, and because Sheu et al. (2004) and Carton and Adam (2003) found, in their studies on national differences in MRP adoption of companies using MRP systems provided by global suppliers, that MRP implementations and adoption can be more difficult in Europe than in North America. The reason for this is that European companies have a more complex culture, compared to more homogenous standards in the US, which essentially makes a successful implementation of a multi-national MRP system in Europe very difficult. This argument is further supported by the results from Olhager and Selldins (2003) research on Swedish manufacturing firms. They found that because of cultural and national issues, Swedish companies, unlike companies in other countries such as the US, generally prefer European and Swedish MRP suppliers to large global suppliers. This suggests that culture differences in Norway and the US can affect the implementation of a US developed MRP system in a Norwegian based company.

# 8.2 The influence of culture on the implementation of an MRP system in NOVN

Hofstede developed a cultural dimensions theory consisting of five dimensions that is widely supported for capturing cross-country differences. These dimension are power distance, uncertainty avoidance, individualism, masculinity, and long-term orientation (Albers-Miller & Gelb, 1996; Baskerville, 2003; Fernandez et al., 1996; Lynn & Gelb, 1996; Soares et al., 2007; Yoo et al., 2011). Even though all of these dimensions are important to take into consideration, the authors of this master thesis believes three dimensions are relevant for explaining in what way cultural differences can affect the implementation of an MRP system. These three dimensions are power distance, individualism, and uncertainty avoidance, and the following sections discusses their relation to the US, Norway, and MRP implementation.

#### Power distance

Compared to the US, Norway scores relatively low on the dimension of power distance. This means that power is decentralised in Norway and that employees in Norwegian companies strive for equalisation of power and opportunity for everyone, whilst the members of a hierarchical society such as the US, accept that inequality of power and wealth will continue to grow within the society (Cebuc & Iosif, 2008; The Hofstede Centre, 2014). This suggests that MRP implementation can be affected by a company's structure due to the culture of each company, which characterises its way of doing business. If the MRP system requires that NOVN must change its organisation structure it can lead to an incorrect application of data, which increases NOVN's employees resistance to change. This can lead to the MRP system being configured with the existing system, thus making it difficult for the new system to offer the improvement it is meant to offer. When implementing a new system, it is therefore vital to

have support from top management to achieve a successful implementation. Examples of support could be delegating and relying on the abilities of employees on lower levels, which can often be seen in Norwegian companies.

#### Uncertainty avoidance

Uncertainty avoidance can be seen as the extent to which people feel threatened by uncertainty and ambiguity and try to avoid these situations (Hofstede & Hofstede, 2014; Soares et al., 2007; Yoo et al., 2011). Even though both Norway and the US fall in the middle of the range with their scores on uncertainty avoidance, companies within these countries can still experience resistance towards change as most people dislikes change (Cebuc & Iosif, 2008; Hofstede, 2011; The Hofstede Centre, 2014). This suggests that the implementation of the MRP system can be affected by NOVN's employees' willingness to change. In regards to a new information system, this resistance among NOVN's employees can occur as they might believe that the new system will change the way they work and increase their workload, or that their job will be redundant as the new system automates their work. NOVN can reduce this resistance by informing its employees, as early as possible, about the change of PPC system, how the new system will affect the employee's jobs, as well as how the implementation is going to be conducted. This way the employees may feel more involved in the implementation process. However, it is important to be aware of that resistance to change and lack of adaptability are issues that are difficult to manage and that will require much energy, time and money from top management if an implementation is going to be successful.

#### Individualism

Both Norway and the US can be seen as individualistic societies. Being an individualistic society means that individuality and individual rights are crucial, that personal opinions are valued and expressed, and that self-interest comes before the interest of a group (Cebuc & Iosif, 2008; The Hofstede Centre, 2014). Individualism can be reflected in how the implementation of the MRP system can be affected by individual or group interests, including the selection of the implementation team and training activities. The choice of implementation team is a critical phase, and NOVN must not underestimate the importance of involving the right employees in this process. Another critical phase for achieving a successful implementation is the training phase. It is important that NOVN's employees are trained in how the new system works and how it can be fully exploited. This training process is, among

others, dependent on the employee's perception of the new system, how open the employees are to change, and the ease with which the new system suits NOVN.

The above discussions show how different companies have a desire to retain their own strong cultural identity and way of doing things. For instance, NOV in the US carries out most of the work themselves, whilst NOVN outsource much of its construction work to countries with cheaper labour, such as Poland. Another example of cultural differences between Norway and the US is how engineers work: in Norway the engineers have a desire to use their education and be innovative and not to do the same thing over and over again, whilst engineers in the US do what is expected of them and do not try to make improvements all the time.<sup>3</sup>

Ignoring the differences in culture will not necessarily lead to failure of implementation, but awareness of these cultural differences and preferences will definitely improve the MRP implementation. This implies that a one-size-fits-all approach is not likely to be successful, which means that the MRP system must be adapted to fit NOVN's culture.

# 8.3 Other aspects that NOVN must take into consideration when implementing the MRP system

In addition to the arguments mentioned above, NOVN must also consider the importance of communication and sharing of information on all levels as lack of communication towards and from top management can generate misunderstandings that can lead to delays in the implementation process. Good communication, as well as global collaboration, will often lead to faster progress.

Several authors argue that positive organisational factors, in particular top management, in large extent contribute to success when adopting new technology (Bowonder et al., 1994; Hwang et al., 2004). The reason for this is that top management can stimulate change by communicating and reinforcing values for the organisation. This is possible as top management has the greatest capacity and status in influencing the behaviour of other members and in resource allocation in the organisation (Thong et al., 1996). With strong support from top management, the resources that are necessary for conducting the adoption and implementation of the new system can be mobilised when needed. This prioritisation can

<sup>&</sup>lt;sup>3</sup> Source of information is collected from the conducted interviews.

encourage the entire organisation to focus on the implementation, as well as motivate employees to adapt to the new system (DeLone, 1981). This demonstrates that the greater the top management support is in NOVN, the easier it will be for NOVN to overcome the difficulties and complexities faced with the implementation and adoption of the new MRP system.

When implementing and adopting systems of large magnitude, which NOVN is about to do, it is important to remember to not rush the vendor selection process. In fact, this process should be given a great deal of attention. Most vendors are helpful in providing complete products and assisting in the adoption process, but it is still dependent on expertise of internal in-house employees. The most important reason for this is to make sure that the new system is customised and integrated into the company, and thus is compatible with the existing systems of the company (Michell & Fitzgerald, 1997). As NOVN already have selected their vendor of the new MRP system, and because they already are working on customising the new system to suit their needs and existing systems, this subject will not be further elaborated.

Another important factor for NOVN to consider is to what degree the new MRP system is consisting with the existing vision, past experiences, and needs of NOVN. The compatibility of the new MRP system and its users, as well as the operational procedures of NOVN, can influence the adoption of the new system as well as slow down the adoption process and discourage the end users (Thong, 1999). It is therefore particularly important for NOVN to consider as many scenarios as possible. This can be done by including the different departments in the adoption and implementation process and listen to their requirements and needs for being able to do their job. This is especially important for NOVN as they are about to implement an American developed program into a Norwegian based company. Because of this, the authors of this thesis recommend NOVN to take their time with the implementation process, listen to the employees concerns and needs, and do their best to make sure these concerns and needs are taken into consideration. It is also recommendable that NOVN specify to its employees that when the new system is implemented it will take some trial and error until the system works optimally. NOVN should also; to the extent it is possible, implement the new system in relevant departments as a trial run, before the system is implemented into the entire organisation. This trial run should cover at least an entire calendar year. By doing this, it will be possible to discover issues and weaknesses that must be resolved before the system is implemented in the entire organisation.

The above discussions emphasises that the early stage of implementation is the most critical. At this point the foundation of the implementation is laid down, and it is from these foundations that the system is customised to meet the requirements of NOVN. The training of the systems users, which here is NOVN's employees, is also essential because the employees are going to manage the system on a day-to-day basis. This means that the employees must be made fully aware of the MRP systems capabilities to exploit the system. The authors of this master thesis therefore recommend NOVN to give its employees thorough training of the new MRP system, as this will increase the chances for achieving a successful implementation.

## **9** CONCLUSION

This final chapter summarises the research findings of the master thesis. A concise answer to each of the research questions as well as a summary of the research findings in relation to the overall research objective is articulated in Section 9.1, along with a clear statement of the extent to which findings can be generalised in Section 9.2. Finally, the scope and limitations of the research conducted, and the avenues for further research are identified in Section 9.3 and 9.4, respectively.

## 9.1 Answers to the research questions

In this research the authors have investigated if implementation of an MRP system supports the operations strategy of an ETO company by conducting a case study of an ETO company that is about to implement an MRP system; NOVN. This case study has resulted in a better understanding of the research objective, as well as the specific research questions that the objective was broken down into. Therefore, this master thesis contributes new knowledge to the research literature and for other researchers. Moreover, the findings of this master thesis can be utilised as a guideline for future research in similar research areas.

The literature review that was carried out was first and foremost used to read up on relevant theory that was applicable for the research. Further work with the literature provided an insight into what previous research had been conducted on the research topic of this thesis. The authors found that a relatively little amount of research has been carried out on the impact of an MRP system in an ETO company. This resulted in both a challenging and time-consuming process of finding applicable literature.

The case study conducted required a profound understanding of NOVN's engineering processes. The case study was therefore time-consuming as the NOVN's processes are both complex and to some extent difficult to comprehend. To make it possible for the case study to be conducted, it was important that NOVN provided the necessary facilities for the research to be conducted in a satisfactorily manner.

## 9.1.1 RQ1: What is the impact of an ATO based PPC system on an ETO company?

The findings of RQ1 show that the roughneck unit's ATO operations strategy, in comparison to the current ETO operations strategy, was more effective in meeting customer requirements in terms of dependability and value added. However, the findings also suggests that the ATO based PPC system did not support the overall objective of the ATO operations strategy, which was to improve delivery lead-time and product quality. The main reasons for the ATO based PPC systems' performance has been identified to be miscommunication of goals and lack of top management support.

# 9.1.2 RQ2: How can an MRP system be used to support the PPC needs of an ETO company?

The findings of RQ2 show that there is a clear misalignment between the decision support provided by an MRP system and the decision support required by an ETO company. The product-, market- and process characteristics of an ETO company are too much of a constraining factor to the MRP system, which may lead to reduced competitiveness. Furthermore, the research suggests that organisational factors, such as education level of employees, company size and culture have significant impacts on implementation of an MRP system.

The result of this research supports the view of Bertrand and Muntslag (1993) and Little et al. (2000), which is that an MRP system does not perform satisfactorily in a company with an ETO operations strategy. The research findings suggests that the main challenge of implementing an MRP system in NOVN is MRP system nervousness, which could reduce the company's flexibility and hence its competitiveness. NOVN should also learn from past experiences with implementing PPC systems and avoid miscommunication of goals of the new MRP system, as well as ensuring a dedicated top management visible for the organisation as a whole. Culture and its potential impact on the implementation of an MRP system in NOVN have also been identified as an important factor affecting successful MRP implementation. Furthermore, findings suggest that additional elements, such as a supportive top management, early involvement of the employees and adequate training, will increase the chances of achieving a successful MRP implementation.

## 9.1.3 Summary of research findings in relation to the overall research objective of the thesis

Recall that the overall research objective is to investigate if implementation of an MRP system supports the operations strategy of an ETO company. Chapter 2, which makes up the theoretical background of the thesis, argues that achieving a strategic fit between a company's operations strategy and its product-, market- and production characteristics are a key determinant to increase the company's performance and achieve long-term success. Furthermore, the strong link between the market requirements and operations strategy concerns the choice of a PPC system. How to assess a PPC system's impact on a company's performance is further explained by using five performance measures and other subjective measures. A review of the PPC system MRP establishes the system's main functions, as well as its advantages and drawbacks. The main objective of the MRP system is to provide the right part at the right time and to meet schedules for completed products. This suggests that the MRP system's main advantage is its rescheduling function, while nervousness is its main drawback. Finally, a literature review on the use of an MRP system in an ETO environment is undertaken and unveils that there is a lack of literature on the MRP and ETO relationship. The small amounts of literature that have touched upon this relationship have concluded that an MRP system will not function optimally in an ETO company. Chapter 3 presents the overall research objective of the thesis and its specific research questions. It also explains the different variables used to test the formulated hypotheses. Chapter 4 expands Chapter 3, explaining how research methods are applied in the study and linking these with the research questions. The case study conducted is a mixed method research using both descriptive and exploratory research design in terms of quasi experiment and interviews. Collected data was analysed using ANOVA and Scheffe's test, as well as content analysis. Chapter 5 introduces the case study company NOVN and its product-, market- and production characteristics. This chapter also elaborates on the ATO operations strategy of the roughneck unit, and its ATO based PPC system. A relationship between the PPC systems impact on a company's performance and the five performance measures and other subjective measures was established in the theoretical background. Chapter 6 investigates what impact the ATO based PPC system had on the roughneck unit and the reason for this impact. Findings suggests that the ATO based PPC system did not support the main objective of the ATO operations strategy mainly due to miscommunication of goals and lack of top management support. Chapter 7 investigates the applicability of a MRP system within NOVN based on the main features of the MRP system defined in the theoretical background. Findings suggest that there is a clear misalignment between the decision support provided by an MRP system and the decision support required by an ETO company. Lastly, based on findings from the interviews, additional elements to consider when implementing an MRP system has been discussed in Chapter 8, with a particular focus on culture.

## 9.2 Generalizability

While this master thesis is grounded in the MRP and ETO literature, it focuses on the oil and gas industry. The case study and its related interviews are all conducted within the oil and gas industry. Therefore this thesis only focuses on the potential to generalise to the ETO oil and gas industry, rather than the wider ETO sector.

## 9.3 Scope and limitations

PPC and MRP is a large research area involving many aspects and variables on different planning levels, and while validity concerns were addressed in Section 4.5, there are inevitable limitations associated with the research and the methods adopted to conduct the research. Because the scope of this research is to investigate if implementation of an MRP system supports the operations strategy of an ETO company, this research brought with it the following limitations:

- Because the thesis has a broad research question, and because the case study only included one unit within a large company, the results must be viewed with some caution in regards to generalizability.
- The conducted research is based on a non-sophisticated manual system that is not found in any literature. However, the authors, in collaboration with their contact person at NOVN, have come to the conclusion that the system can be applied as a measure for the purpose of the research.
- Although parts of this research attempted to answer the research questions specific to the use of an ATO based PPC system, the thesis used traditional MRP theory as its basis. This meant that the focus would be on basic functionalities of a traditional MRP system, and not the ATO based PPC system. Because of this the results must be viewed with some caution. To the extent that these systems are found to adhere to MRP theory the results of this research can be applied to the ATO based PPC system.
- As the main competitive advantage of NOVN lies in its enquiry-, design- and engineering stages, which are also considered to be the company's main bottlenecks,

this research only focused on analysing the implementation of an MRP system in respect to the PPC needs at these stages.

- Due to NOVN's limited use of the ATO based PPC system; the long-term effects of an MRP system were not studied.
- To conduct the research requested by NOVN, the company granted the authors of this thesis access to facilities, key informants and data required to carry out the research. A contract has regulated the details of the use and publication of company data and information. Therefore, only certain information provided by NOVN is published in this research.
- The presented findings, based on the qualitative data gathered in this master thesis, are the authors' own interpretation and evaluation of the facts and research evidence. Interpretations must therefore be viewed with some caution.

## 9.4 Further research

This research has added structure to a relatively unchartered research field. Hence, a range of future research avenues can be identified. Constraint relating scope, time, finance and access prevented the uptake of these areas. Without these constraints, the authors of this thesis is confident that the results of a more extensive research would provide a more comprehensive framework that will give NOVN a much better understanding of how the implementation of an MRP system can support their ETO operations strategy.

For future research on this topic, the authors suggest that other researchers with larger resources and more available time go more in depth and use multiple cases and units. A more elaborated research would be possible to achieve by looking into other units within NOVN that also have used the ATO based PPC system. This would provide more data, thus making it possible to compare results from several units.

It would be of interest to perform the same research in different industries, such as shipbuilding, tool making and construction. From such research it would be possible to gain a more complete picture of the phenomenon and to generalise findings more than the authors were able to do in this research.

It would also be of interest to use other research methodologies, such as a survey and time series experiment, to see if they provide the same results, thus confirming the results received in this research. As well as make a more thorough analysis of some of the factors that may have had an impact on the results of this research. An example could be to investigate to what extent the financial crisis from 2007 had an impact on the decision to abandon the ATO based PPC system used by the roughneck unit.

As some of the informants indicated that culture and cultural differences could have a major impact on the implementation of an MRP system, it would be of interest to perform an extensive research on this topic and the obstacles it can lead to.

Another possibility would be to extend the research to also include how an ETO company can mitigate MRP system nervousness.

This overview of further research topics are far from complete, however it gives an idea of how extensive the research possibilities of ETO and MRP are. It also provides an indication of that MRP systems are here for the long haul and that their impact on different operations strategies will need to be studied on an on-going basis.

## REFERENCES

- Aghazadeh, S. M. (2003). MRP contributes to a copmany's profitability. *Assembly Automation*, 23(3), 257-265. doi:10.1108/01445150310486521
- Albers-Miller, N. D., & Gelb, B. D. (1996). Business advertising appeals as a mirror of cultural dimensions: a study of eleven countries. *Journal of Advertising*, 25(4), 57-70. doi:10.1080/00913367.1996.10673512
- Anderson, C. R., & Zeithaml, C. P. (1984). Stage of the product life cycle, business strategy, and business performance. Academy of Management, 27(1), 5-24. doi:10.2307/255954
- Anderson, J. C., Schroeder, R. G., & Ballard, G. (1984). Getting results from your MRP system. *Business Horizons*, 27(3), 57-64. doi:10.1016/0007-6813(84)90028-4
- Ang, J. S. K., Sum, C. C., & Chung, W. F. (1995). Critical success factors in implementing MRP and government assistance: a Singapore context. *Information & Management*, 29(2), 63-70. doi:10.1016/0378-7206(95)00017-Q
- Aslan, B., Stevenson, M., & Hendry, L. C. (2012). Enterprise resource planning systems: an assessment of applicability to make-to-order companies. *Computers in Industry*, 63(7), 692-705. doi:10.1016/j.compind.2012.05.003
- Avraham, S., & Karni, R. (2013). ERP The Dynamics of Supply Chain and Process Management doi:10.1007/978-0-387-74526-8
- Barnes, D. (2008). *Operations management An international perspective* (1 ed.). London: South-Western Cengage Learning.
- Baskerville, R. F. (2003). Hofstede never studied culture. *Accounting, Organizations and Society*, 28(1), 1-14. doi:10.1016/S0361-3682(01)00048-4
- Bendoly, E., & Jacobs, F. R. (2006). ERP architectual/operational alignment for orderprocessing performance. *International Journal of Operations & Production Management*, 24(1), 99-117. doi:10.1108/01443570410511013
- Berry, W., & Cooper, M. (1999). Manufacturing Flexibility: Methods for Measuring the Impact of Product Variety on Perfromance in Process Industries. *International Journal* of Operations & Production Management, 17(2), 163-178. <u>http://fisher.osu.edu/supplements/10/1470/Berry\_1999\_Summary\_Manufacturing\_Fle\_xibility.pdf</u>
- Berry, W. L., & Terry, T. (1992). Linking Systems to Strategy. International Journal of Operations & Production Management, 12(10), 3-13. http://search.proquest.com/docview/232338401?accountid=45259
- Bertrand, J. W. M., & Muntslag, D. R. (1993). Production control in engineer-to-order firms. *International Journal of Production Economics*, *30*(1), 3-22. doi:10.1016/0925-5273(93)90077-X
- Bhattacharyya, D. K. (2006). *Research Methodology* (2 ed.): Excel Books.
- Blackburn, J. D., Kropp, D. H., & Millen, R. A. (1986). A comparison of strategies to dampen nervousness in MRP systems. *Management Science*, 32(4), 413-429. doi:10.1287/mnsc.32.4.413
- Bowonder, B., Miyake, T., & Linstone, H. A. (1994). The Japanese institutional mechanisms for industrial growth: A systems perspective—Part I. *Technological Forecasting and Social Change*, 47(2), 229-254. doi:10.1016/0040-1625(94)90030-2
- Boyer, K., & Verma, R. (2010). *Operations & Supply Chain Management for the 21st Century* (1 ed.). Mason, OH: South-Western Cengage Learning.
- Boyer, K. K., & McDermott, C. (1999). Strategic consensus in operations strategy. *Journal of Operations Management*, 17(3), 289-305. doi:10.1016/S0272-6963(98)00042-4

- Bozarth, C., & Chapman, S. (1996). A contingency view of time-based competition for manufacturers. *International Journal of Operations & Production Management*, 16(6), 56-67. doi:10.1108/01443579610119090
- Buonanno, G., Faverio, P., Pigni, F., Ravarini, A., Sciuto, D., & Tagliavini, M. (2005). Factors affecting ERP system asoption: A comparative analysis between SMEs and large companies. *Journal of Enterprise Information Management*, 18(4), 384-426. doi:10.1108/17410390510609572
- Burns, O. M., Turnipseed, D., & Riggs, W. E. (1991). Critical success factors in manufacturing resource planning implementation. *International Journal of Operations* & Production Management, 11(4), 5-19. doi:10.1108/01443579110136221
- Callerman, T. E., & Heyl, J. E. (1986). A model for material requirements planning implementation. *International Journal of Operations & Production Management*, 6(5), 30-37. doi:10.1108/eb054778
- Carlson, R. C., Jucker, J. V., & Kropp, D. H. (1979). Less nervous MRP systems: a dynamic economic lot-sizing approach. *Management Science*, *25*(8), 754-761. doi:10.1287/mnsc.25.8.754
- Carton, F., & Adam, F. (2003). Analysing the impact of enterprise resource planning systems roll-outs in multi-national companies. *Electronic Journal of Information Systems Evaluation*, 6(2), 21-32. <u>http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.71.9047&rep=rep1&type=p</u> df
- Cebuc, G., & Iosif, L. C. (2008). Cultural diversity in the European business environment EU-US comparison. *Romanian Economic and Business Review*, 3(1), 18-33. http://www.rebe.rau.ro/REBE 3 1.pdf - page=18
- Cheng, T. C. E., & Musaphir, H. (1996). Theory and practice of manufacturing strategy. *International Journal of Production Research*, 34(5), 1243-1259. <u>http://web.b.ebscohost.com/ehost/pdfviewer/pdfviewer?sid=67eb2964-dfdb-4190-8fb1-fcca403eb7ae%40sessionmgr115&vid=2&hid=124</u>
- Childerhouse, P., & Towill, D. (2000). Engineering supply chains to match customer requirements. *Logistics Information Management*, *13*(6), 337-346. doi:10.1108/09576050010355635
- Chopra, S., & Meindl, P. (2007). *Supply chain management strategy, planning, & operation* (3 ed.). New Jersey: Pearson Education Inc.
- Cook, T. D., & Campbell, D. T. (1979). *Quasi-Experimentation: Design & Analysis Issues for Field Settings* (1 ed.). U.S.A.: Houghton Mifflin Company.
- Cutler, T. R. Special orders: Engineer-to-order challenge for the industrial engineer. Industrial Engineer. <u>http://www.trcutlerinc.com/industrial\_engineer\_may.pdf</u>
- D'Avino, M., De Simone, V., & Schiraldi, M. M. (2013). Revised MRP for reducing inventory level and smoothin order releases: a case in manufacturing industry. *Production Planning & Control: The Management of Operations*, 16(1). doi:10.1080/09537287.2013.764579
- Davison, R. (2002). Cultural complications of ERP. *Communication of the ACM*, 45(7), 109-111. <u>http://web.b.ebscohost.com/ehost/pdfviewer/pdfviewer?sid=65a857d2-fb9d-</u> 41c1-8aa2-955dd4329bdc%40sessionmgr115&vid=2&hid=112
- Deep, A., Guttridge, P., Dani, S., & Burns, N. (2004). Investigating factors affecting ERP selection in made-to-order SME sector. *Journal of Manufacturing Technology Management*, 19(4), 430-446. doi:10.1108/17410380810869905
- DeLone, W. H. (1981). Firm size and the characteristics of computer use. *MIS Quarterly*, 5(4), 65-77. doi:10.2307/249328
- Doyle, C. (2011). Dictionary of Marketing doi:10.1093/acref/9780199590230.001.0001

- Elfving, J. A., Tommelein, I. D., & Ballard, G. (2005). Consequences of competetive bidding in project-based production. *Journal of Purchasing and Supply Management*, 11(4), 173-181. doi:10.1016/j.pursup.2005.12.001
- Ferdows, K., & Meyer, A. D. (1990). Lasting improvements in manufacturing performance: in search of a new theory. *Journal of Operations Management*, 9(2), 168-184. doi:10.1016/0272-6963(90)90094-T
- Fernandez, D. R., Carlson, D. S., Stepina, L. P., & Nicholson, J. D. (1996). Hofstede's country classification 25 years later. *The Journal of Social Psychology*, 137(1), 43-54. doi:10.1080/00224549709595412
- Fjose, S., Holmen, R. B., & Gulbrandsen, M. U. (2013). *MENON Report* Vol. 34. *Value Added and Ripple Effects of National Oilwell Varco Norway* Retrieved from http://menon.no/a/ringvirkninger-av-national-oilwell-varco-norways-aktivitet
- Flynn, B. B., & Flynn, E. J. (2004). An exploratory study of the nature of cumulative capabilities. *Journal of Operations Management*, 22(5), 439-457. doi:10.1016/j.jom.2004.03.002
- Gianesi, I. G. N. (1998). Implementing manufacturing strategy through strategic production planning. *International Journal of Operations & Production Management*, 18(3), 286-299. doi:10.1108/01443579810196499
- Gosling, J. (2011). Flexibility strategies for engineer-to-order construction supply chains. (Doctoral thesis, Cardiff University). Retrieved from http://orca.cf.ac.uk/54426/1/U584527.pdf
- Gosling, J., Naim, M., & Towill, D. (2013). A supply chain flexibility framework for engineer-to-order systems. *Production Planning & Control: The Management of Operations*, 24(7), 552-566. doi:10.1080/09537287.2012.659843
- Gosling, J., & Naim, M. M. (2009). Engineer-to-order supply chain management: A literature review and research agenda. *International Journal of Production Economics*, 122(2), 741-754. doi:10.1016/j.ijpe.2009.07.002
- Grabenstetter, D. H., & Usher, J. M. (2013). Determining job complexity in an enginner to order environment for due date estimation using a proper framework. *International Journal of Production Research*, *51*(19), 5728-5740. doi:10.1080/00207543.2013.787169
- Grant, R. M. (2003). Strategic planning in a turbulent environment: evidence from the oil majors. *Strategic Management Journal*, 24(6), 491-517. doi:10.1002/smj.314
- Gunasekaran, A., Patel, C., & McGaughey, R. E. (2004). A framework for supply chain performance measurement. *International Journal of Production Economics*, 87(3), 333-347. doi:10.1016/j.ijpe.2003.08.003
- Hallgren, M., & Olhager, J. (2006). Differentiating manufacturing focus. *International Journal of Production Research*, 44(18-19), 3863-3878. doi:10.1080/00207540600702290
- Hameri, A. P. (1997). Project management in a long-term and global one-of-a-kind project. *International Journal of Project Management*, 15(3), 151-157. doi:10.1016/S0263-7863(96)00051-8
- Hayes, R. H. (1985). Strategic planning-forward in reverse? *Harvard Business Review*, 63(1), 67-77. doi:10.1080/00913367.1996.10673512
- Heizer, J., & Render, B. (1988). *Production and Operations management* (1 ed.). Massachusettes: Allyn and Bacon Inc.
- Hicks, C., & Braiden, P. M. (2001). Computer-aided production management issues in the engineer-to-order production of complex capital goods explored using a simulation approach. *Strategic Finance*, 83(3), 30-34. http://web.a.ebscohost.com/ehost/detail?sid=e4e3f95b-9bb7-4caa-83c2-

<u>7b8ee640dfd6%40sessionmgr4003&vid=1&hid=4112&bdata=JnNpdGU9ZWhvc3Qt</u> bG12ZQ%3d%3d - db=bth&AN=5058354

- Hicks, C., McGovern, T., & Earl, C. F. (2000). Supply chain management: A strategic issue in engineer to order manufacturing. *International Journal of Production Economics*, 65(2), 179-190. doi:10.1016/S0925-5273(99)00026-2
- Hicks, C., McGovern, T., & Earl, C. F. (2001). A Typology of UK Engineer-to-Order Companies. International Journal of Logistics Research and Application: A Leading Journal of Supply Chain Management, 4(1), 43-56. doi:10.1080/13675560110038068
- Hickson, D., Pugh, D. S., & Pheysey, D. C. (1969). Operations technology and organization structure: an empirical reappraisal. *Administrative Science Quarterly*, 14(3), 378-397. <u>http://web.b.ebscohost.com/ehost/pdfviewer/pdfviewer?sid=ccbab0f0-bbf4-4877ab23-059c3b3532cd%40sessionmgr113&vid=2&hid=123</u>
- Hill, A., & Hill, T. (2009). *Manufacturing Operations Strategy* (3 ed.): Palgrave Macmillan.
- Ho, C. J. (1989). Evaluating the impact of operating environments on MRP system nervousness. *International Journal of Production Research*, 27(7), 1115-1135. doi:10.1080/00207548908942611
- Ho, C. J. (2002). Evaluating dampening effects of alternative lot-sizing rules to reduce MRP system nervousness. *International Journal of Production Research*, 40(11), 2633-2652. doi:10.1080/00207540210134489
- Ho, C. J., & Ho, S. J. K. (1999). Evaluating the effectiveness of using lot-sizing rules to cope with MRP system nervousness. *Production Planning & Control: The Management of Operations*, 10(2), 150-161. doi:10.1080/095372899233307
- Ho, C. J., & Ireland, T. C. (1993). A diagnostic analysis of the impact of forecast errors on production planning via MRP system nervousness. *Production Planning & Control: The Management of Operations*, 4(4), 311-322. doi:10.1080/09537289308919453
- Ho, C. J., Law, W. K., & Rampal, R. (1995). Uncertainty-dampening methods for reducing MRP system nervousness. *International Journal of Production Research*, 33(2), 483-496. doi:10.1080/00207549508930161
- Ho, C. Y. (2007). Exploring the compatibility of dampening procedures and lot-sizing rules in MRP systems under uncertain operating environments. *International Journal of Production Research*, 46(18), 5097-5120. doi:10.1080/00207540701338887
- Hofstede, G. (2011). Dimensionalizing cultures: the Hofstede model in context. *Online Reading in Psychology and Culture, 2*(1), 1-26. doi:10.9707/2307-0919.1014
- Hofstede, G., & Hofstede, J. G. (2014). Culture. Retrieved 09.05.2014, from <u>http://geerthofstede.nl/culture.aspx</u>
- Hong, K. K., & Kim, Y. G. (2002). The critical success factors for ERP implementation: an organizational fit perspective. *Information & Management*, 40(1), 25-40. doi:10.1016/S0378-7206(01)00134-3
- Huang, G. Q., Yee, W. Y., & Mak, K. L. (2001). Development of a web-based system for engineering change management. *Robotics and Computer-Integrated Manufacturing*, 17(3), 255-267. doi:10.1016/S0736-5845(00)00058-2
- Hwang, H. G., Ku, C. Y., Yen, D. C., & Cheng, C. C. (2004). Critical factors influencing the adoption of data warehouse technology: a study of the banking industry in Taiwan. *Decision Support Systems*, *37*(1), 1-21. doi:10.1016/S0167-9236(02)00191-4
- Ip, W. H. (1998). Manufacturing integration strategy using MRP II and RTMs: a case study in South China. *Integrated Manufacturing System*, 9(1), 41-49. doi:10.1108/09576069810196832
- Islam, S., Rahman, M., Saha, R. K., & Saifuddoha, A. (2013). Development of Material Requirements Planning (MRP) Software with C Language. *Global Journal of Computer Science and Technology Software & Data Engineering*, 13(3).

 $https://globaljournals.org/GJCST\_Volume 13/3-Development-of-Material-Requirements.pdf$ 

- Jin, G., & Thomson, V. (2003). A new framework for MRP systems to be effective in engineered-to-order environments. *Robotics and Computer-Integrated Manufacturing*, 19(6), 533-541. doi:10.1016/S0736-5845(03)00063-2
- Jodlbauer, H., Olhager, J., & Schonberger, R. J. (2012). Modelling value. *Contributions to Management Science*, 1(1), 37-46. doi:10.1007/978-3-7908-2747-7
- Jonsson, P. (2008). *Logistics and supply chain management* (1 ed.). New York: McGraw-Hill Education.
- Keung, K. W., Ip, W. H., & Chan, C. Y. (2001). An enhanced MPS solution for FMS using GAs. *Integrated Manufacturing System*, 12(5), 351-359. doi:10.1108/EUM000000005712
- Kim, Y., & Lee, J. (1993). Manufacturing strategy and production systems: an integrated framework. *Journal of Operations Management*, 11(1), 3-15. doi:10.1016/0272-6963(93)90029-O
- King, N., & Horrocks, C. (2010). *Interviews in Qualitative Research* (1 ed.). London: SAGE Publications.
- Klassen, R. D., & Menor, L. J. (2006). *Cases in Operations Management*: SAGE Publications, Inc.
- Klugh, H. E. (1986). *Statistics: the essentials for research* (3 ed.). Hillsdale, Nj: Lawrence Erlbaum Associates.
- Koh, S. C., Jones, M. H., Saad, S. M., Arunachalam, S., & Gunasekaran, A. (2000). Measuring uncertainties in MRP environments. *Logistics Information Management*, 13(3), 177-183. doi:10.1108/09576050010326574
- Konijnendijk, P. A. (1994). Coordinating marketing and manufacturing in ETO companies. International Journal of Production Economics, 37(1), 19-26. doi:10.1016/0925-5273(94)90004-3
- Krumbholz, M., & Maiden, N. (2001). The implementation of enterprise resource planning packages in different organisational and national cultures. *Information Systems*, 26(3), 185-204. doi:10.1016/S0306-4379(01)00016-3
- Laukkanen, S., Sarpola, S., & Hallikainen, P. (2007). Enterprise size matters: objectives and constraints of ERP adoption. *Journal of Enterprise Information Management*, 20(3), 319-334. doi:10.1108/17410390710740763
- Lee, T. S., & Adam Jr, E. E. (1986). Forecasting error evaluation in material requirements planning (MRP) production-inventory systems. *Management Science*, 32(9), 1186-1205. doi:10.1287/mnsc.32.9.1186
- Lee, Y. Y., Kramer, B. A., & Hwang, C. L. (1990). Part-period balancing with uncertainty: a fuzzy sets theory approach. *International Journal of Production Research*, 28(10), 1771-1778. doi:10.1080/00207549008942832
- Leidner, D. E., & Kayworth, T. (2006). Review: a review of culture in information systems research: toward a theory of information technology culture conflict. *MIS Quarterly*, *30*(2), 357-399. <u>http://www.jstor.org/stable/25148735</u>
- Little, D., Rollins, R., Peck, M., & Porter, K. (2000). Integrated planning and scheduling in the engineer-to-order sector. *International Journal of Computer Integrated Manufacturing*, *13*(6), 545-554. doi:10.1080/09511920050195977
- Lynn, M., & Gelb, B. D. (1996). Identifying innovative national markets for technical consumer goods. *INternational Marketing Review*, 13(6), 43-57. doi:10.1108/02651339610151917
- Mabert, V. A. (2007). The early road to material requirements planning. *Journal of Operations Management, 25*(2), 346–356. doi:10.1016/j.jom.2006.04.002

- Mabert, V. A., Soni, A., & Venkataramanan, M. A. (2003). The impact of organization size on enterprise resource planning (ERP) implementations in the US manufacturing sector. *Omega*, *31*(3), 235-246. doi:10.1016/S0305-0483(03)00022-7
- Marshall, L. F. (1997). What is the Right Supply Chain for Your Product. *Harward Business Review*, 75(2), 105-116. http://www.computingscience.nl/docs/vakken/scm/Fisher.pdf
- Martínez-Olvera, C., & Shunk, D. (2006). Comprehensive framework for the development of a supply chain strategy. *International Journal of Production Research*, 44(21), 4511-4528. doi:10.1080/00207540600621698
- McGovern, T., Hicks, C., & Earl, C. F. (1999). Modelling supply chain management processes in engineer-to-order companies. *International Journal of Logistics Research* and Applications: A Leading Journal of Supply Chain Management, 2(2), 147-159. doi:10.1080/13675569908901578
- Michell, V., & Fitzgerald, G. (1997). The IT outsourcing market-place: vendors and their selection. *Journal of Information Technology*, *12*(1), 223-237. doi:10.1080/026839697345080
- Minifie, J. R., & Davis, R. A. (1990). Interaction effects on MRP nervousness. *International Journal of Production Research*, 28(1), 173-183. doi:10.1080/00207549008942691
- Morabito, V., Pace, S., & Previtali, P. (2005). ERP marketing and Italian SMEs. *European Management Journal*, 23(5), 590-598. doi:10.1016/j.emj.2005.09.014
- Moustakis, V. (2000). *Material Requirements Planning MRP* Retrieved from <u>http://www.adi.pt/docs/innoregio\_MRP-en.pdf</u>
- Muscatello, J. R., Small, M. H., & Chen, I. J. (2003). Implementing enterprise resource planning (ERP) systems in small and midsize manufacturing firms. *International Journal of Operations & Production Management*, 23(8), 850-871. doi:10.1108/01443570310486329
- Nagare, H. D. (2007). *Machine shop production planning*. (Master thesis, Welingkar Institute if Management Development & Research). Retrieved from https://<u>http://www.welingkaronline.org/DLP/OPERATIONS (REVISED) -</u> Machineshop Production and planning-HEMANT NAGARE.pdf
- Narasimhan, R., & Melnyk, S. A. (1984). Assessing the transient impact of lot sizing rules following MRP implementation. *International Journal of Production Research*, 22(5), 759-772. doi:10.1080/00207548408942497
- Naylor, J. B., Naim, M. M., & Berry, D. (1999). Leagility: Integrating the lean and agile manufacturing paradigms in the total supply chain. *International Journal of Production Economics*, 62(1-2), 107-118. doi:10.1016/S0925-5273(98)00223-0
- NOV. (2013). Iron Roughnecks [brochure]. Houston, TX: NOVN.
- NOV. (2014a). About NOV. Retrieved 29.01.2014, from <u>http://www.nov.com/aboutnov.aspx?id=5225&linkidentifier=id&itemid=5225</u>
- NOV. (2014b). National Oilwell Varco announces fourth quarter and full year 2013 earnings. *Internal company report*.
- NOV. (2014c). Segments. Retrieved 25.10.2014, from https://http://www.nov.com/Segments.aspx
- NOV Inc. (2006). *Annual report 2005*. Washington, D.C.: United States Securities and Exchange Commission.
- NOV Inc. (2010). Annual report 2009. Washington, D.C.: United States Securities and Exchange Commission.
- NOV Milestones [video clip]. (2014). from http://nov.com/aboutnov.aspx?id=5225&linkidentifier=id&itemid=5225
- NOVN. (2008). NOV BOD meeting.
- NOVN. (2011). National Oilwell Varco Rig Solutions Norway. Internal company report.

NOVN. (2012). The NOV Norway organisation. Internal company report.

NOVN. (2013). We thrive in deep waters [brochure]. Kristiansand: NOVN.

- Olhager, J. (2003a). Linking manufacturing strategy and production planning and control. *Production Planning & Control: The Management of Operations*, 14(6), 485-486. doi:10.1080/09537280310001621949
- Olhager, J. (2003b). Strategic positioning of the order penetration point. *International Journal* of Production Economics, 85(3), 319-329. doi:10.1016/S0925-5273(03)00119-1
- Olhager, J. (2012). The role of decoupling points in value chain management. *Management Science*, 1(1), 37-47. http://download.springer.com/static/pdf/746/chp%3A10.1007%2F978-3-7908-2747-
- 7 2.pdf?auth66=1410451638 e2afce07e52612bc79b46a2303db28ff&ext=.pdf Olhager, J., & Selldin, E. (2003). Enterprise resource planning survey of Swedish
- manufacturing firms. European Journal of Operational Research, 146(2), 365-373. doi:10.1016/S0377-2217(02)00555-6
- Olsen, K. A., & Sætre, P. (2007). IT for niche companies: is an ERP system the solution. Information Systems Journal, 17(1), 37-58. doi:10.1111/j.1365-2575.2006.00229.x
- Olson, D. L., Chae, B., & Sheu, C. (2005). Issues in multinational ERP implementation. International Journal of Services and Operations Management, 1(1), 7-21. http://inderscience.metapress.com/content/xpq6t13p2an63xnq/
- Ortega, C. H., Garrido-Vega, P., & Dominguez Machuca, J. A. (2012). Analysis of interaction fit between manufacturing strategy and technology management and its impact on performance. *International Journal of Operations & Production Management*, 32(8), 958-981. doi:10.1108/01443571211253146
- Penlesky, R. J., Berry, W. L., & Wemmerlöw, U. (1989). Open order due date maintenance in MRP systems. *Management Science*, *35*(5), 571-584. doi:10.1287/mnsc.35.5.571
- Petroni, A. (2002). Critical factors of MRP implementation in small and medium-sized firms. *International Journal of Operations & Production Management*, 22(3), 329-348. doi:10.1108/01443570210417623
- Petroni, A., & Rizzi, A. (2001). Antecedents of MRP adoption in small and medium-sized firms. *Benchmarking: An International Journal*, 8(2), 144-156. doi:10.1108/14635770110389852
- Platts, K. W., Mills, J. F., Bourne, M. C., Neely, A. D., Richards, A. H., & Gregory, M. J. (1998). Testing manufacturing strategy formulation processes. *International Journal* of Production Economics, 56-57(1), 517-523. doi:10.1016/S0925-5273(97)00134-5
- Porter, M. E. (1983). The Technological Dimension of Competitive Strategy. *Research on Technological Innovation, Management and Policy*, 7(1).
- Porter, M. E. (1985). The Competitive Advantage: Creating and Sustaining Superior Performance. *The Free Press, 1*(1).
- Ptak, C., & Smith, S. (2008). Beyond MRP. Retrieved 20.02.2014, from <u>http://www.prinetsol.com/pdf/tocico2008/Actively Synchronized Replenishment</u> <u>White Paper.pdf</u>
- Rahim, A. R. A., & Baksh, M. S. N. (2003). The need for a new product development framework for engineer-to-order products. *European Journal of Innovation Management*, 6(3), 182-196. doi:10.1108/14601060310486253
- Ram, B., Naghshineh-Pour, M. R., & Yu, X. (2006). Material requirements planning with flexible bills-of-material. *International Journal of Production Research*, 44(2), 399-415. doi:10.1080/00207540500251505
- Reinertsen, R. Ø. (2013, 14.10.). GE eller NOV? *Fædrelandsvennen*. Retrieved from <u>http://www.fvn.no/okonomi/bedrift/GE-eller-NOV-2488549.html - .U3sDK1h\_vaY</u>

- Reinertsen, R. Ø. (2014, 22.04.). Alle tiders sørlandsrekord fra NOV. *Fædrelandsvennen*. Retrieved from <u>http://www.fvn.no/okonomi/Alle-tiders-sorlandsrekord-fra-NOV-2592977.html</u>
- Rink, D. R. (1979). Product life cycle research: a literature review. *Journal of Business Research*, 7(3), 219-242. doi:10.1016/0148-2963(79)90030-4
- Roy, R. N. (2005). *A Modern Approach to Operations Management* (2 ed.). New Dehli: New Age International (P) Ltd., Publishers.
- Samadhi, T. M. A. A., & Hoang, K. (1995). Shared computer-integrated manufacturing for various types of production environment. *International Journal of Operations & Production Management*, 15(5), 95-108. doi:10.1108/01443579510083695
- Samaranayake, P., Lewis, G. S., Woxwold, E. R. A., & Toncich, D. (2002). Development of engineering structures for scheduling and control of aircraft maintenance. *International Journal of Operations & Production Management*, 22(8), 843-867. doi:10.1108/01443570210436172
- Saunders, M., Philip, L., & Thornhill, A. (2009). *Research methods for business students* (5 ed.). Essex: Pearson Education Limited.
- Schroeder, R. G., Anderson, J. C., Tupy, S. E., & White, E. M. (1981). A study of MRP benefits and costs. *Journal of Operations Management*, 2(1), 1-9. doi:http://dx.doi.org/10.1016/0272-6963(81)90031-0
- Sheu, C., Chae, B., & Yang, C. L. (2004). National differences and ERP implementation: issues and challanges. *Omega*, 32(5), 361-371. doi:10.1016/j.omega.2004.02.001
- Silvestro, R., Fitzgerald, L., Johnston, R., & Voss, C. (1992). Towards a lassification of service processes. *International Journal of Service Industry Management*, 3(3), 62-75. doi:10.1108/09564239210015175
- Skinner, W. (2007). Manufacturing strategy: the story of its evolution. *Journal of Operations Management*, 25(2), 328-335. doi:10.1016/j.jom.2006.10.008
- Slack, N., Chambers, S., & Johnston, R. (2004). *Operations Management* (4 ed.): Pearson Education Limited.
- Soares, A. M., Farhangmehr, M., & Shoham, A. (2007). Hofstede's dimensions of culture in international marketing studies. *Journal of Business Research*, 60(3), 277-284. doi:10.1016/j.jbusres.2006.10.018
- Soh, C., Kien, S. S., & Tay-Yap, J. (2000). Cultural fits and misfits: is ERP a universal solution? *Communication of the ACM*, 43(4), 47-51. <u>http://web.a.ebscohost.com/ehost/pdfviewer/pdfviewer?sid=5d5c3579-36c6-4753a154-eb79b7d3269e%40sessionmgr4005&vid=2&hid=4209</u>
- Sriram, P. K., Alfnes, E., & Arica, E. (2013). A concept for project manufacturing planning and control for engineer-to-order companies. Advances in Production Management Systems. Competitive Manufacturing for Innovative Products and Services, 398(1), 699-706. doi:10.1007/978-3-642-40361-3\_89
- Stevenson, M., Hendry, L. C., & Kingsman, B. G. (2005). A review of production planning and control: the applicability of key concepts to the make-to-order industry. *International Journal of Production Research*, 43(5), 869-898. doi:10.1080/0020754042000298520
- Sum, C. C., & Ng, K. K. (2014). MRP Implementation. Retrieved 25.02, 2014, from http://www.springerreference.com/docs/html/chapterdbid/6940.html
- Sum, C. C., & Yang, K. K. (1993). A study on manufacturing resource planning (MRP II) practices in Singapore. *Omega*, 21(2), 187-197. doi:10.1016/0305-0483(93)90051-L
- Sum, C. C., Yang, K. K., Ang, J. S. K., & Quek, S. A. (1995). An analysis of material requirements planning (MRP) benefits using alternating conditional expectation
(ACE). Journal of Operations Management, 13(1), 35-58. doi:10.1016/0272-6963(95)00005-D

- Swamidass, P. M., & Newell, W. T. (1987). Manufacturing strategy, environmental uncertainty and performance: a path analytic model. *Management Science*, 33(4), 509-524. doi:10.1287/mnsc.33.4.509
- Tang, Y., & Y.C., P. (2010). The review of Misfit Issues between ERP Principles and Organisations. 2010 International Conference on Education and Management Technology, 1(1). <u>http://ieeexplore.ieee.org/stamp/stamp.jsp?tp=&arnumber=5657545</u>
- The Hofstede Centre. (2014). The Hofstede Centre. from <u>http://geert-hofstede.com/norway.html</u>
- Thompson, I., Cox, A., & Anderson, L. (1998). Contracting strategies for the project environment. *European Journal of Purchasing & Supply Management, 4*(1), 31-41. doi:10.1016/S0969-7012(98)00005-7
- Thong, J. Y. L. (1999). An integrated model of information systems adoption in small businesses. *Journal of Management Information Systems*, 15(4), 187-214. <u>http://ssrn.com/abstract=1977980</u>
- Thong, J. Y. L., Yap, C. S., & Raman, K. S. (1996). Top management support, external expertise and information systems implementation in small businesses. *Information Systems Research*, 7(2), 248-267. doi:10.1287/isre.7.2.248
- van Donk, D. P. (2003). The decoupling point in the food processing industries. *International Journal of Production Economics*, 69(3), 297-306. doi:10.1016/S0925-5273(00)00035-9
- Vokurka, R. J., Zank, G. M., & Lund III, C. M. (2002). Improving competitiveness through supply chain management: a cumulative improvement approach. *Competitiveness Review*, 12(1), 14-25. doi:10.1108/eb046431
- Vollmann, T. E., Berry, W. L., Whybark, D. C., & Jacobs, F. R. (2005). *Manufacturing Planning and Control for Supply Chain Management* (5 ed.): McGraw Hill.
- Vrijhoef, R., & Koskela, L. (2000). The four roles of supply chain management in construction. *European Journal of Purchasing & Supply Management*, 6(3-4), 169-178. doi:10.1016/S0969-7012(00)00013-7
- Wadhwa, S., Bhoon, K. S., & Chan, F. T. S. (2006). Postponement strategies through business process redesign in automative manufacturing: Knowledge innovation. *Industrial Management & Data Systems*, 106(3), 307-326. doi:10.1108/02635570610653470
- Wagner, S. M., Ruyken, P. T., & Erhun, F. (2012). The link between supply chain fit and financial performance of the firm. *International Journal of Operations Management*, 30(2), 340-353. <u>http://ac.els-cdn.com/S0272696312000022/1-s2.0-S0272696312000022-main.pdf?\_tid=58f13ae0-e5d9-11e3-bb05-00000aacb35f&acdnat=1401220921\_45e073c1ac3f0a9e001f96930a23425f</u>
- Wemmerlöw, U. (1989). The behaviour of lot-sizing procedures in the presence of forecast errors. *Journal of Operations Management*, 8(1), 37-47. doi:10.1016/S0272-6963(89)80004-X
- Wemmerlöw, U., & Whybark, D. C. (1984). Lot-sizing under uncertainty in a rollin schedule environment. *International Journal of Production Research*, 22(3), 467-484. doi:10.1080/00207548408942467
- White, E. M., Anderson, J. C., Schroeder, R. G., & Tupy, S. E. (1982). A study of the MRP implementation process. *Journal of Operations Management*, 2(3), 145-153. doi:10.1016/0272-6963(82)90002-X
- White, G. P. (1996). A meta-analysis model of manufacturing capabilities. *Journal of Operations Management*, 14(4), 315-331. doi:10.1016/S0272-6963(96)00096-4

- Wikner, J., & Rudberg, M. (2005). Integrating production and engineering perspectives on the customer order decoupling point. *International Journal of Operations & Production Management*, 25(7), 623-641. doi:10.1108/01443570510605072
- Wong, C. M., & Kleiner, B. H. (2001). Fundamentals of material requirements planning. *Management Research News*, 24(3-4), 9-12. doi:10.1108/01409170110782540
- Wortmann, H. (1995). Comparison of information systems for engineer-to-order and make-tostock situations. *Computer in Industry*, 26(3), 261-271. doi:10.1016/0166-3615(95)00047-8
- Xu, X., & Liang, Z. (2011). CODP positioning based on extension superiority evaluation model. *Electronic and Mechanical Engineering and Information Technology*, 8(1), 4041-4047. doi:10.1109/EMEIT.2011.6023940
- Yang, B., & Burns, N. (2003). Implications of postponement for the supply chain. International Journal of Production Research, 41(9), 2075-2090. doi:10.1080/00207544031000077284
- Yin, R. K. (2009). Case Study Research: Design and Methods (4 ed.): SAGE Publications Inc.
- Yin, R. K. (2012). Applications of Case Study Research (3 ed.): SAGE Publications Inc.
- Yoo, B., Donthu, N., & Lenartowicz, T. (2011). Measuring Hofstede's five dimensions of cultural values at the individual level: development and validation of CVSCALE. *Journal of International Consumer Marketing*, 23(3-4), 193-210. doi:10.1080/08961530.2011.578059
- Zijm, W. H. M. (2000). Towards intelligent manufacturing planning and control systems. *OR-Spektrum*, 22(3), 313-345. <u>http://link.springer.com/article/10.1007/s002919900032</u>
- Zikmund, W. G., Babin, B. J., Carr, J. C., & Griffin, M. (2013). *Business Research Methods* (9 ed.). Canada: South-Western Cengage Learning.

# APPENDIX

# **Interview guide**

## **General questions**

## NOV

- What do you think is the most important performance measure of NOVN?
- What do you think are qualifying and order winning factors?
- Where do you believe the majority of NOV's products are in the PLC?

# Roughneck

#### Individual characteristics

- How long have you worked at NOVN?
- What is your current position and responsibilities?
- What was your role during the implementation of the previously used serial production system?
- What were your specific responsibilities?
- How long were you in this position?
- How many employees worked in the roughneck unit during the use of the system?
- Can you explain what the roughneck unit do?

### MRP (serial production system)

- What is your knowledge on serial production?
- What do you think are the main advantages and disadvantages of a serial production system?

### Motivation and support for implementing MRP

- Who took the initiative to implement the serial production system?
- Why was it decided to introduce the serial production system in this unit?
- What was the goal?
- What did you think about the goals clarity and relevance?

#### The evaluation of the implementation

#### Evaluation of system efficiency and user-friendliness

- In what way did you experience that it was easier to gain access to necessary information?
- In what way did you experience that the information you received from the system was reliable?
- In what way did you experience that employees became more efficient after the implementation?
- What kind of feedback did you receive from the employees regarding the system?
- In what way did you experience that the employees were reluctant to adapt to the system?
- In what way did you experience increased customer satisfaction after the implementation?

#### Evaluation of the results

- What were the results of the implementation, and how were they measured?
- Did you, in your opinion, experience better competitiveness through the usage of the system?
- Why was it decided to shut down the system?
- In your opinion, do you believe one could have achieved better results if you got to continue the production under the serial production system over a longer period of time, than what was the case?
- Describe how you feel the system satisfied the main goals?

## **Oracle (The new MRP system)**

#### Motivation and support for the implementation of Oracle

- Are you aware that Oracle will be implemented in NOVN?
- Based on what you know, who took the initiative to implement Oracle?
- Based on what you know, what are the main goals NOVN want to achieve by implementing Oracle?

#### Planning the implementation of Oracle

- What specific results do you expect from this implementation?

#### The use of MRP

#### BOM / MPS / inventory

- In what way do you believe that Oracle will change the BOM's accuracy?
- In what way do you believe that Oracle will change the MPS's accuracy?
- In what way do you believe that Oracle will change inventory's accuracy?
- To what extent do you believe that departments must be computer based to support inventory, BOM and MPS, and why?

#### **Evaluation of implementation**

#### Evaluation of system efficiency and user friendliness

- In what ways can Oracle help make your job easier?
- Do you expect that the introduction of Oracle will contribute to easier decision making, and in what ways?
- In what way do you expect that the introduction of Oracle will make it easier to access necessary information?
- In what way do you expect that the information provided by Oracle is reliable?
- Do you expect that the employees will do a more effective and efficient job after the implementation?
- Do you expect that employees will be reluctant to adapt to Oracle?
- Do you expect increased customer satisfaction after implementation?

#### Evaluation of the results

- How do you think the result of the implementation of Oracle should be measured?
- Do you expect better competitiveness using Oracle?

#### **Evaluation of Oracle**

- In your opinion, what are the strengths and weaknesses of Oracle?
- In your opinion, in what way is Oracle compatible with the operations and production processes that are performed within NOVN?
- In your opinion, how must Oracle be tailored to fit NOVN?

- What do you believe are the critical success factors for the implementation of Oracle and why?
- In regards to the introduction of Oracle, what do you believe are the most significant implementation issues?

<u>Lastly</u>

What do you believe are the biggest differences between NOVN and the rest of NOV?